

Received September 15, 2021, accepted September 26, 2021, date of publication September 29, 2021, date of current version October 7, 2021.

Digital Object Identifier 10.1109/ACCESS.2021.3116259

Wind-to-Hydrogen Production Potential for Selected Sites in Pakistan

JAMEEL AHMAD^{®1}, MUHAMMAD IMRAN², SYED FAROOQ ALI³, MUHAMMAD ADNAN^{®1}, SYED REHAN ASHRAF^{®4}, ZAWAR HUSSAIN⁵, AND MUHAMMAD SHOAIB¹

¹Department of Computer Science, School of Systems and Technology, University of Management and Technology (UMT), Lahore 54700, Pakistan
 ²Department of Mechanical, Biomedical and Design Engineering, College of Engineering and Physical Sciences, Aston University, Birmingham B4 7ET, U.K.
 ³Department of Software Engineering, School of Systems and Technology, University of Management and Technology (UMT), Lahore 54700, Pakistan
 ⁴Department of Industrial Engineering, School of Engineering, University of Management and Technology (UMT), Lahore 54700, Pakistan

⁵Department of Electrical Engineering, School of Engineering, University of Management and Technology (UMT), Lahore 54700, Pakistan

Corresponding author: Jameel Ahmad (jameel.ahmad@umt.edu.pk)

This work was supported by the University of Management and Technology (UMT), Lahore, Pakistan.

ABSTRACT This paper provides a comprehensive assessment of wind energy potential for Hydrogen production under local conditions of Pakistan from the design, development and practical implementation perspectives. Simulations were performed for three sites—Bahawalpur, Sanghar and Gwadar using actual wind speed site data, recorded between 2016 and 2018, at intervals of 10 minute. For the selected sites, wind resource assessment was performed using the Weibull distribution function parameters, energy and wind-power density calculations at hub heights of 20m, 40m, 60m and 80m of the wind turbines. It was observed that Sanghar is the most suitable site for wind-to-Hydrogen production potential with power and energy density of 305.86 W/m^2 and 2665.81 kWh/m^2 , respectively. From the implementation perspective, the Nordex N90/2500 wind turbine at an 80m hub height was found to be beneficial for Sanghar with a cost of energy of 35.21\$/MWh (0.035\$/kWh). The cost of Hydrogen using an electrolyzer for 7-year long-term investment was 2.29 k\$/ton using Nordex N90/2500 turbine. Based on the available power density and land area, a general scheme for production of Hydrogen using electrolysis can be implemented with possibility of installation and commissioning of wind farms.

INDEX TERMS Economic assessment, hydrogen production, Weibull distribution, wind energy.

I. INTRODUCTION

Today world is facing the global challenge of climate change and temperature rise. The quest of finding a lasting solution to pollution-free clean air environment has led to the formulation of various policies and agreements like the Kyoto Protocol and Paris Agreement by all stake holders across the globe [1], [2]. Due to technological advancement in most developed countries, the demand for energy to sustain these buoyant economies such as extraction of gas Hydrogen and liquid Hydrogen from wind power and their techno-economic analysis, offshore oil and gas exploration has seen an upward trajectory in the last few decades [3]–[5]. Unlike developing countries, these giant economies have exploited various medium of energy generation to meet their demand during peak hours. In developing countries, the gap between the energy demand and supply is increasing because

The associate editor coordinating the review of this manuscript and approving it for publication was Ton Duc $Do^{\textcircled{D}}$.

of the rapid growth in population, depletion of fossil fuel resources, high cost of oil and gas imports, the increase in energy costs and therefore trying to enhance energy efficiency and doing exergy, exergoeconomic and exergoenvironmental analysis [6], [7]. These observations are seen particularly in Bangladesh, India, Pakistan, Iran and some countries of Africa and Asia. An increasing trend has been observed in the utilization of green Hydrogen [8] in developing countries to cope with the energy crisis and preserving healthy environment with less Carbon pollution [9].

Depending upon the geographical location and local conditions, the most common resources of renewable energy in these countries include solar, wind, hydro and biomass along with sustainable development of Hydrogen from these resources. Feasibility study of six resources of renewable energy namely solar, municipal solid waste, biomass, micro-hydro, and geothermal for Hydrogen production was evaluated using fuzzy analytical hierarchy process, Fuzzy Delphi, and environmental data envelopment analysis (DEA) in [10]. It was concluded that production of Hydrogen can be made using wind energy. From available resources of renewable energy, wind energy is considered an integral part of various energy scenarios with the current worldwide installed power capacity of approximately 733.3 GW in 2020 [11]. In year 2020, renewable energy kept on growing exponentially. Moreover, solar and wind capacity increased 50% compared to any previous expansion [12]. Weather conditions play a vital role in the generation of wind power. These two parameters, wind speed and output power of wind turbines, have a cubic bond association and hence, the precise wind resource estimation for any site is considered significantly important [13], [14].

In [15] authors have provided current status of wind energy projects in general, but specifically the wind corridor of Jamshoro city in Sindh province of Pakistan and estimated parameters of Weibull distribution for Jamshoro at 50, 30 and 10m hub-height and determined wind energy potential. An overall status of Pakistan wind power plant projects is also provided. A recent study is done by authors in [16]. In this study authors have provided a through outlook on wind speed pattern of various sites, on wind energy deployment, its impact on environment and various technical challenges in adopting this source of renewable energy. Weibull distribution parameters for wind energy assessment for six locations (Gwadar, Jiwani, Karachi, Ormara, Pasni and Sonmiani Bay) in coastal belt of Sindh province in Pakistan is carried out in [17] and annual energy production, economic feasibility analysis, payback period estimation and cost of energy is determined. Different wind turbine technologies for power generation are proposed in the study. Similar studies were done in [18] for Hawke's Bay, in [19] for Hyderabad, in [20] for Babaurband, in [21] for Gharo and in [22] for Karachi sites situated in wind corridor of Sindh province of Pakistan. The land in Jhimpir area which is 127km away from Karachi, while traveling on Hyderabad motorway M-9 via nearby city Noori Abad, has been allocated initially for ten wind farms for power generation. Benchmark wind speeds for this site are estimated in [23]. The terrain of Jhimpir is hard and rocky and elevation of 40-50 m from sea level. Many companies later on got license for power generation and started installing their wind farms of 50 MW power output. The list of the companies working in Jhimpir is provided in [15]. All of these studies did not include production of Hydrogen from wind power. However, the installation of Vensys 62 was done at Thatta, Jhimpir, and operated by Zorlu Enerji Pakistan Ltd [24]. As can be seen in Table 6 in [24], wind resource capacity has a potential of 2.354×105 MW, that translates to production of around 45,000 tonnes of Hydrogen (considering a conversion value of 53 KWh per kg of Hydrogen).

Despite the increasing focus of research on electrical energy storage technologies, only a few industry scale storage facilities have been installed worldwide [25]. In this context, increasing research has focused on the environmental benefits of generating electrolytic Hydrogen using wind power as comparable to other methods of Hydrogen production [26]. The life-cycle assessment of wind-based electrolysis showed that 1 kg of CO_2 is emitted for the production of 1 kg of Hydrogen [27]. Based on this life-cycle assessment, wind-based electrolysis presents lower CO_2 emissions than other renewable energy systems [28], [29]. Mirza et al. [30] presented the complete road map to Hydrogen economy in Pakistan. The integration of Hydrogen economy with the society will lead to technological advancements in many fields, including transportation, power generation using renewable energy sources, and agriculture. For Hydrogen production through electrolysis, solar energy and wind energy are the most suitable renewable energy resources for electricity generation. However, the latter is more effective because of the lower levelized cost [31]. Olateju et al. [32] assessed the production of electrolytic Hydrogen for application in the oil sand bitumen upgrading industry in Western Canada. The results showed that the Hydrogen production cost is 7.84 USD/kg and is uncompetitive with respect to conventional fossil fuel Hydrogen pathways. Alavi et al. [33] carried out a feasibility assessment of wind energy for production of Hydrogen in Baluchestan and Sistan in Iran. The wind-power potential and Hydrogen production were assessed at five sites, and a maximum Hydrogen production of 39.82 tons per annum was observed. However, this study did not consider the technical and economic aspects in detail.

Al-Sharafi et al. [34] analyzed the Hydrogen and electricity production from solar and wind power in Saudi Arabia. The optimization results showed that the minimum levelized cost of energy is 0.609 USD/kWh, and the corresponding cost of the Hydrogen production is 43.1 USD/kg. Mostafaeipour et al. [35] investigated the wind energy capacity for Hydrogen production for car fuel in the Fars province of Abadeh city, Iran. For the selected site, the maximum annual electrical energy was 575.53 MWh and could meet the Hydrogen load of 22 cars per week. Rezaei-Shouroki et al. [36] selected 13 cities of Fars for the development of a wind farm for producing Hydrogen utilizing wind-power density estimates. It was concluded that Izad Khast city has the highest potential of Hydrogen production with an average of 21.9 tons/year using a 900-kW wind turbine. Ashrafi et al. [37] investigated the wind potential in detail using different variants of Weibull parameters for Hydrogen production in Iran. For the estimation of Weibull parameters, various methods were employed, namely: empirical method of Lysen, the power density method [38] and the standard deviation method.

Hydrogen production was depicted using Geographic Information Science maps. However, the study did not include economic aspects. In 2018, Gökçek and Kale [39] proposed an optimization-based scenario for Hydrogen fuel produced by PV-wind hybrid systems in Izmir-Cesme, Turkey. National Aeronautics and Space Administration surface meteorology and solar energy database were utilized for solar and wind potential assessment. The levelized cost of Hydrogen production was found to be 7.526-7.866 USD/kg under different system configurations.

Ayodele and Munda [40] investigated the economic feasibility for the production of green Hydrogen using wind energy and water electrolysis at five different locations in South Africa. The simulation results showed that the maximum Hydrogen production was 226.82 metric-tons with an annual storable compressed Hydrogen content of 598 m3. Sensitivity analysis concluded that the wind speed significantly affected Hydrogen production and its associated viability. Chade et al. [41] analyzed the wind turbinediesel-electrolyzer system for Hydrogen production by considering the example of Grimsey Island and showed that the proposed system was economically feasible with a payback period of 4 years for the optimal system configuration. Douak and Settou [42] presented methods for production of Hydrogen using wind energy in Algeria. Wind resource assessment was performed using the Weibull probability distribution function. The results showed that the minimum cost of Hydrogen production was 1.214 USD/kg. A comprehensive review of Hydrogen production from renewable energy from the economic perspective is presented in Table 1.

Thus, numerous studies have focused on the potential of Hydrogen production via electrolysis using wind energy. However, from deployment perspective, it is essential to consider all techno-economic aspects of wind power utilization for wind-to-Hydrogen production and realization. Moreover, there is a clear research gap, in previous works, between the actual wind resource assessment, technical and economic aspects of Hydrogen production and implementation strategies. This study considers the case study of Pakistan and estimates the Hydrogen production based on the potential of wind energy resources of three selected sites. The novelty of this study is that for the first time, to the best of our knowledge, a study has focused on estimate of Hydrogen production from wind energy resource assessment and most economically relevant wind turbine technologies with promising payback for Sanghar, Gwadar and Bahawalpur sites. Although, there is an existing study for the production of renewable Hydrogen using wind energy in Sindh, Pakistan that includes various sites, namely: Talhar, Keti Bandar, Baghan, Gharo, DHA Karachi, Jamshoro, Nooriabad, and Golarchi [54], however, the authors did not find any published work of this kind for our selected sites. This study will act as catalyst for oil-dependent nations having wind energy resources for Energy and Hydrogen production. The summary of our contribution for selected sites of Pakistan in this perspective is highlighted as follows;

- Wind resource assessment through Weibull Distribution
- Calculation of power and energy densities for these sites.
- Selecting the best choice of wind turbines
- Calculation of levelized cost of energy and payback period.
- Estimation of cost of Hydrogen production using windto-Hydrogen energy conversion system
- Exploration of possible pathways for distribution of Hydrogen from production sites

The rest of the paper is organized as follows. Section II provides the conceptual framework for current research work. Section III contains data analysis of the sites and subsequent wind resource assessment. Section IV outlines techno-economic assessment of integrating wind turbines for power generation their payback period. In Section V a model for Hydrogen production is provided and cost analysis is carried out with Hydrogen delivery options for a given transport radius. Section VI gives a good account of discussion on results and paper is concluded in Section VII.

II. RESEARCH FRAMEWORK

Knowledge regarding the characteristics of wind regimes of the selected site is important to provide an overview of the economic viability of the locations for electric power generation and subsequent utilization for Hydrogen production from water electrolysis. A schematic diagram of the proposed research framework is shown in Fig. 1. In the following sections we will first do a wind resource assessment for electricity generation through selection of the most suitable wind energy conversion system followed by investigation for a Hydrogen production model and then at the end life-cycle cost analysis through break-even analysis.

III. WIND DATA ANALYSIS FOR WIND RESOURCE ASSESSMENT

We analyzed satellite data for the period of 2000 - 2010and existing ground data, and the initial results indicate a good wind regime in the country, as shown in Fig. 2(a) [55]. The selected sites with their geographical details are shown in Table 2 and shown on a GIS map in Fig. 2(b). Temporal wind speed data recorded for the selected sites for two years with 10 min intervals have been preprocessed by converting it into daily and monthly mean wind speeds at heights of 20m, 40m, 60m, and 80m.

Wind speeds for Pakistan are classified into various classes according to the National Renewable Energy Laboratory (NREL) wind map as shown in Fig. 2(a) [55]. Using this classification, it could easily assess the wind regime of different provinces of Pakistan. Because of global growth in the production of power from wind energy, various studies have been conducted in the last two decades with an aim to statistically model the potential of wind and its frequency distribution in many countries. This frequency distribution approach can help simply predict the energy output of a typical wind-energy-conversion system. Various methods have been tested for fitting the measured wind speed data to a target probability distribution function (PDF) for a given location over a period of time, typically on a monthly or yearly basis. However, choosing the best PDF that fits the wind speed distribution is a challenging task.

Some of the most widely adopted distribution functions include the Weibull and Rayleigh distributions to characterize wind speed patterns for a given location under consideration. The two parameter Weibull distribution has been considered in this study because it has been found to fit a wide variety of

TABLE 1. Economic aspects of production of hydrogen.

Ref.	System details	Hydrogen Cost(USD/kg of H_2)
[43]	Directly coupled PV-PEM electrolyzer	7.94
[44]	Combined PV modules and parabolic trough collectors with SOEC electrolyzer	5.23
[45]	Solar thermal power integrated with alkaline electrolyzer	2.89
[45]	PV modules with alkaline electrolyzer	7.22
[46]	Grid assisted PV-PEM electrolyzer	6.1
[47]	Electrolyzer powered by wind turbine	3.1
[48]	Wind turbine and alkaline electrolyzer	2.53
[49]	Electrolyzer powered by wind turbine	8.5
[50]	Electrolyzer powered by wind turbine	4.34
[51]	Electrolyzer powered by wind turbine	3.37
[52]	Wind turbine-electrolyzer-PEM fuel cell	5
[53]	SOEC electrolyzer using geothermal energy for water preheating	1.84
[53]	Geothermal plant with CO_2 as working fluid and alkaline electrolyzer	8.24



FIGURE 1. Conceptual schematic diagram of the wind-Hydrogen plant.

TABLE 2.	Site o	lata and	its	geographical	coordinates.
----------	--------	----------	-----	--------------	--------------

Site	Longitude	Latitude	Province	No of Data Samples	Data Time Period
Bahawalpur	$71^{\circ}41'56.03''E$	$25^{\circ}19'35.97''N$	Punjab	115645	01/04/2016 - 06/30/2018
Sanghar	$69^{\circ}2'15.12''E$	$25^{\circ}48'57.26''N$	Sindh	65965	01/12/2016 - 30/11/2017
Gwadar	$60^{\circ}20'46.95''E$	$25^{\circ}16'47.30''N$	Balochistan	50910	01/12/2016 - 30/11/2017

wind speed data [56], [57]. At a specific wind farm, the available electricity generated by a wind turbine generator system depends on the mean wind speed, its skewness, standard deviation, and kurtosis. Considering the data location, data sample size, goodness of fit test, and statistical inference criteria, one estimation method can demonstrate better performance than other methods [58]. In this study, a comparison of four different estimation methods is proposed for selected sites to estimate the Weibull parameters. The deviation of estimated results from Weibull Distribution from the measured values of wind speed and their effect on annual energy produced by wind turbines is also examined. The yearly energy estimate is further utilized to calculate the cost of electricity per kilo watt hour and production of Hydrogen using water electrolysis. The amount of Hydrogen generated is used for long-term, midterm, and short-term cost of Hydrogen estimate for economic feasibility of the installation of wind turbines at various sites. The research framework is based on the flowchart given in Fig. 3.

A. WIND RESOURCE ASSESSMENT OF SELECTED SITES

The distribution pattern of wind speed data can be investigated using two statistical estimators, namely: skewness¹ and kurtosis.² These statistical expressions are defined as follows [60], [61];

skewness =
$$\frac{1}{T-1} \Sigma_{i=1}^{T} \frac{(v_i - \bar{v})^3}{\sigma^3}$$
(1)

kurtosis =
$$\frac{1}{T-1} \Sigma_{i=1}^{T} \frac{(v_i - \bar{v})^4}{\sigma^4} - 3$$
 (2)

where v_i , \bar{v} , and σ are the wind speed (m/sec) at *ith* observation, mean, and standard deviation of wind speed respectively. *T* is the number of non-zero instances in wind speed data. Wind characteristics for selected sites are listed in Table 3.

B. ESTIMATION OF WEIBULL PARAMETERS

The Weibull PDF [60], f(v) is shown in Eq. (3).

$$f(v) = (\frac{k}{c})(\frac{v}{c})^{k-1} exp(\frac{-v}{c})^k$$
(3)

Here *c* and *k* are scale and shape factor respectively. Several numerical methods can be used to estimate the parameters of the Weibull PDF [62]–[64] using wind speed data. Some of these numerical methods include the graphical method (GM), empirical method (EM), energy pattern factor (EPF) and the maximum likelihood estimate (MLE) [65]–[68].

1) GM

The cumulative distribution function(CDF), F(v) is given by Eq. (4).

$$F(v) = 1 - exp(\frac{-v}{c})^k \tag{4}$$

Taking natural logarithms twice, we get Eq. (5), which is linear in the parameters to be fit. Given the plot of $ln[-ln\{1-F(v_i)\}]$ versus ln(v), c is computed from y-intercept and k equals the slope of the plot.

$$ln[-ln\{1 - F(v_i)\}] = -klnc + klnv_i$$
(5)

2) EM

Justus *et al.* [69] suggested this method, according to which, the empirical k and c are defined by Eq. (6) and (7) respectively. The gamma function is defined by Eq. (8).

$$k = \left(\frac{\sigma}{\bar{v}}\right)^{-1.086} \tag{6}$$

$$c = \left(\frac{\bar{\nu}}{\Gamma(1+\frac{1}{k})}\right) \tag{7}$$

¹Skewness describes the symmetrical characteristic of the data sequence. ²Kurtosis is a measure of whether the data are light-tailed or heavy-tailed with respect to a Gaussian distribution.

$$\Gamma(X) = \int_0^\infty t^{x-1} e^{-t} dt \tag{8}$$

3) EPF

The EPF method was suggested by [70] and is governed by Eq. (9).

$$E_{pf} = \left(\frac{v^3}{\overline{v}^3}\right) \tag{9}$$

k parameter calculated by EPF method is given by Eq. (10) and c parameter is given by Eq. (7).

$$k = 1 + \left(\frac{3.69}{(E_{pf})^2}\right) \tag{10}$$

4) MLE

The fitting of parameters k and c requires numerical iterations. This method uses a likelihood function of wind speed time series data. Parameters k and c are estimated using the following two equations (Eq. (11) and Eq. (12) derived in [71] and also suggested by [72].

$$k = \left(\frac{\sum_{i=1}^{T} v_i^k \ln(v_i)}{\sum_{i=1}^{T} v_i^k} - \frac{\sum_{i=1}^{T} \ln(v_i)}{T}\right)^{-1}$$
(11)
$$c = \left(\frac{1}{T} \sum_{i=1}^{T} v_i^k\right)^{\frac{1}{k}}$$
(12)

C. STATISTICAL ERROR INDICATORS

Statistical error indicators (root mean square error(RMSE), mean absolute percentage error (MAPE), relative root mean square error (RRMSE), and coefficient of correlation are used as error functions for performance analysis [71]. Mathematical formulation of these parameters is provided below, with P_{WB} and P_M being the power density from Weibull function and actual measured data, respectively.

1) RMSE

$$RMSE = \sqrt{\frac{1}{T} \sum_{i=1}^{T} (P_{i,WB} - P_M)^2}$$
(13)

RMSE provides short-term performance of these models. Smaller +ve values indicates the closeness of match between actual data P_M and estimated power density P_{WB} .

2) MAPE

$$MAPE = \frac{1}{T} \sum_{i=1}^{T} |\frac{(P_{i,WB} - P_M)}{P_M}|$$
(14)

3) RRMSE

$$RRMSE = \frac{\sqrt{\frac{1}{T} \sum_{i=1}^{T} (P_{i,WB} - P_M)^2}}{\frac{1}{T} \sum_{i=1}^{T} (P_M)}$$
(15)



FIGURE 2. (a) Wind resource map of Pakistan based on NREL. (b) GIS map of sites (Gwadar, Sanghar and Bahawalpur) [59].



FIGURE 3. Research Methodology.

The range of accepted values of RRMSE to determine the accuracy of our model is given below [72], [73]:

Usually RRMSE $\leq 10\%$ is considered excellent and RRMSE $\geq 30\%$ indicates poor performance.

4) COEFFICIENT OF CORRELATION (R)

$$R = \frac{\sum_{i=1}^{T} (P_{i,M} - P_{i,WB})^2}{\sum_{i=1}^{T} (P_{i,M} - P_{avg,WB})^2}$$
(16)

R can measure the strength of the correlation between the control and simulated values. Higher R values indicate better model performance [74]. The value of R is between 0 and 1.

5) DIEBOLD-MARIANO TEST FOR STATISTICAL ACCURACY

This method is used in inferential statistics to determine if apriori hypothesis is correct by computing a test statistics. The error between the actual data sequence y and prediction model \hat{y}_n^i can be written as follows for two forecasts:

TABLE 3. Statistical characteristics of winds for selected sites.

IEEEAccess

Site	height	mean speed(m/sec)	std.(m/sec)	skewness	kurtosis
Bahawalpur	20m	3.92	2.18	0.80	3.94
-	40m	4.59	2.34	0.70	3.93
	60m	4.60	2.12	0.02	2.00
	80m	5.48	2.93	0.45	2.98
Sanghar	20m	4.99	2.55	0.56	2.80
	40m	5.65	2.74	0.39	2.73
	60m	6.20	2.98	0.23	2.45
	80m	6.51	3.21	0.15	2.24
Gwadar	20m	4.23	2.59	0.72	2.87
	40m	4.61	2.63	0.64	2.93
	60m	4.79	2.67	0.61	2.97
	80m	4.98	2.80	0.58	2.94

 $e_n^i = y - \hat{y}_n^i$, i = 1, 2. We define a loss function $F(e^i)$ to indicate prediction accuracy [75]. In general, the loss function can be defined in two ways, as follows: Square error loss:

$$F(e_n^i) = (e_n^i)^2$$
(17)

Absolute deviation error loss:

$$F(e_n^i) = |e_n^i| \tag{18}$$

The DM statistic is represented using Eq. (19).

$$DM = \frac{\frac{1}{N} \sum_{n=1}^{N} F(e_{n+h}^2) - F(e_{n+h}^1)}{\sqrt{\frac{s^2}{N}}}$$
(19)

 s^2 represents the variance of $F(e_{n+h}^2) - F(e_{n+h}^1)$. Two hypotheses are developed here.

 $H_0: F(e_{n+h}^2) = F(e_{n+h}^1)$ and $H_1: F(e_{n+h}^2) \neq F(e_{n+h}^1).$

Here h is the prediction step size and is taken as 1. The null hypothesis H_0 represents that the prediction performance of the two models are same, that is, the two predictions are not significantly different. In contrast, the other hypothesis H_1 represents a significant difference in the predictive performance of the two models. The level of statistical significance is set to α and the critical value of the test is $Z_{\alpha/2}$. If $|DM| > Z_{\alpha/2}$, the null hypothesis H_0 could be rejected, and if $|DM| \prec Z_{\alpha/2}$, there is not enough evidence to show that a significant difference occurs between the two models. It is further assumed that DM follows a normal distribution with zero mean and unit variance i.e. N(0, 1). We used four estimating models in our current study. We will run the DM test for these methods for their statistical accuracy.

D. WIND POWER DENSITY ESTIMATION

There are two methods to calculate Wind-power density (power per unit area)in $\frac{W}{m^2}$ using either measured wind speed data [76] or using the two parameters estimated for Weibull distribution function [77] at the given site.

1) WIND POWER DENSITY USING MEASURED DATA

$$\frac{P}{A} = \frac{1}{2n} \rho \Sigma_{i=1}^{n} v^{3} = \frac{1}{2} \rho \overline{v^{3}}$$
(20)

where P is power(kW) and A is the swept area of wind turbine blades. ρ is air density taken as 1.225 kg/m³.

2) WIND POWER DENSITY ESTIMATE USING WEIBULL PARAMETERS

The k and c values of Weibull Distribution estimated from the four statistical methods (GM, EM, EPF, and MLE) are used to get an estimate of wind power density.

$$\frac{P}{A} = \frac{1}{2}\rho \int_0^\infty v^3 f(v) dv = \frac{1}{2}\rho c^3 \Gamma(1 + \frac{3}{k})$$
(21)

3) CAPACITY FACTOR AND ANNUAL ENERGY PRODUCTION Capacity factor (CF) [78] as given by Eq. (22) is highly dependent on k and c parameters and rated, cut - in and cut - out wind speeds of selected wind turbine. Most of the practical values for CF lie between 20% and 40% depending on site conditions and the technology of the wind turbine.

$$CF = \frac{exp[-(\frac{v_{cut}-in}{c})^{k}] - exp[-(\frac{v_{r}}{c})^{k}]}{(\frac{v_{r}}{c})^{k} - (\frac{v_{cut}-in}{c})^{k}} - exp[-(\frac{v_{cut}-out}{c})^{k}]$$
(22)

Here, v_{cut-in} is the cut-in wind speed, v_r is the rated wind speed, and $v_{cut-out}$ is the cut-out wind speed of the selected wind turbine. The annual energy production, E_{out} for each wind turbine can be calculated using its rated power, P_r , and capacity factor, CF.

$$E_{out} = CF.P_r.8760 \tag{23}$$

IV. ECONOMIC ASSESSMENT

A. WIND TURBINE SELECTION FOR PROPOSED SYSTEM

After careful wind resource assessment, we have a good estimate of how much power can be generated from wind energy. The next phase is to select wind turbines to achieve this goal. Therefore a thorough analysis is done for the most suitable wind turbines from top list of manufacturers. The prices for wind turbines are declining because of a competitive and ever-increasing energy market and advancements in technology. The generation cost of electricity (USD/kWh) is heavily dependent on geo-political factors of the country and current tariffs enforced by regulatory authorities. Various costs that are involved in this process are capital cost,

investment cost, operation and maintenance cost. Capital cost is mainly affected by interest rate and period of loan. Further expenditures of the wind energy conversion project involve prevailing prices for land acquisition, installation, planning, licensing and connection to grid. The operation and maintenance costs include the sum of operation and maintenance cost of each subsystem. This usually involves cost of repair, insurance and administration of spare parts. An in-depth economic analysis is carried out regarding wind turbine selection and finally three wind turbines i.e. DeWind D4/48, DeWind D6, and Nordex N90/2500 are recommended in this work. The wind speed and corresponding power characteristics of the selected wind turbines are listed in Table 4. A certain portion of generated wind power can be fed to utility grid to meet the energy demand of the region whereas a fixed share of remaining power can be consumed by the electrolyzer for Hydrogen production. Various topologies based on power electronics modules to control active and reactive power, voltage and harmonics, dynamic economic dispatch and wind farm layout are possible [79], [80].

Nordex N90/2500 wind turbine having double-fed induction generator (DFIG) technology is designed for name plate capacity of 2500 kW at generation voltage of 660 V with operating frequency of 50 Hz or 60 Hz. These turbines have good performance in variable wind speed regions. The DeWind D4/48 and D6 turbine models have rated powers of 600 kW and 1250 kW, respectively. at 40m and 60m hub heights, a 690-V generation voltage and DFIG design. DFIG technology utilizes rotor-side and grid-side power electronics based controllers for maximum energy capture under variable wind speeds with a tight harmonic control at the output. The characteristics of these wind turbines are compared in detail in [73].

B. COST OF ENERGY

Improvements in technology, regional manufacturing processes, and competitive supply-chains have led to a continuous reduction in the wind turbine prices. According to the data given in a report of the International Renewable Energy Agency (IRENA) 2021, global weighted-average total installed cost and average turbine prices are steadily falling since 2010. However there is an increased trend in capacity factor. Chinese wind turbine prices have fallen drastically for projects commissioned in 2020 [81]. The cost trend according to IRENA report is shown in Fig. 4.

In order to estimate the energy purchase cost (EPC), the National Electric Power Regulatory Authority (NEPRA) of Pakistan has considered many different reports published around globe such as report by IRENA [81], Bloomberg and other sources for wind energy project tariff determination. It finalized that average wind turbine prices in different parts of the world were below USD 0.70 million per MW in 2019 [82]. These reports also forecast the gradual reduction in turbine prices in coming years. Many other factors were considered in estimating EPC besides wind turbine costs

However in its report it argued that the cases of different countries can not exactly be mapped to Pakistan because of different market trends, local manufacturing condition, tariff regimes, performance evaluation and economic factors of other countries. The global trend in decrease of wind turbine prices and corresponding decline in EPC are attributed to competition among turbine manufacturers. Turbines are manufactured with different rotor diameters, hub heights and power generation capacity of turbine etc. Installation costs also vary due to different wind farm layouts, cost of the land with varying terrains, transportation cost, labor cost and level of target performance desired etc. Considering the international trends of reduction in capital cost for onshore wind energy projects and incorporating local geo-political factors for a developing country like Pakistan, NEPRA approved recently (August 18, 2020) cost of approximately USD 50 million per 50 MW for Iran-Pak Wind Power (Pvt.) Ltd (IPWPPL) wind project at Deh Kohistan, District Thatta, Sindh [82]. The details of various project cost components such as EPC, project development cost (PDC), insurance during construction (IDC), financing and fee charges, Karachi Inter-bank Offered Rate (KIBOR) and London Inter-bank Offered Rate (LIBOR) used, can be found in the IPWPPL NEPRA report. The Authority also carefully examined previously determined costs for various wind power projects to approve the new costs in their final decision aligned with the global market trends.

such as local manufacturing cost, cost of land and labor etc.

As per the NEPRA, the approved cost of a wind projects in Pakistan, taking the example of IPWPPL and IRENA report [81], we considered wind power project cost as 1 million USD/MW (USD 1000/kW) [82]–[85] to take into account the local site and country specific conditions. Hence, to estimate the investment cost C_{inv} in the current study, we considered 1000 USD for each kW of nominal capacity of the wind turbine. The present value cost (PVC) or the cost of the kilowatt-hour (kWh) of energy was evaluated based on the following assumptions:

- 1) The investment $\cos (C_{inv})$ consists of the price of wind turbines plus the cost of civil works, installation, land rent, and the connection cables to the grid and is taken as 1 million USD/MW.
- 2) Operation, maintenance and repair costs (C_{omr}), which are related to the general inflation rate, were considered to be 3.5% of the investment cost.
- According to Pakistan's economic indicators given by Trading Economics, Karachi interbank interest rate (KIBOR) (I) and inflation rate (r) were considered to be 10.5% and 9.5%, as of November 2020.
- 4) The wind turbine lifetime (n) is assumed to be 20 years.
- 5) A scrap value (S) of 10% of the investment cost (machine and civil work costs) was chosen.

With these assumptions, the cost of energy per kWh can be calculated using the total energy output, E_{out} , from the wind turbine for its lifetime and PVC from [85], [86] using

Turbine model	Rated	$v_{cut-in}(m/sec)$	$v_r(m/sec)$	$v_{cut-out}(m/sec)$	Hub Height(m)	Rotor	$C_{inv}(USD)$
	Power(kW)					Diameter(m)	
Dewind D4/48	600	2.5	11.5	23	40	48	600000
Dewind D6	1250	3	13	23	60	64.3	1250000
Nordex	2500	3	13	25	80	90	2500000
N90/2500							

TABLE 4. Power and wind speed characteristics of selected wind turbines.



Source: IRENA Renewable Cost Database.

FIGURE 4. Wind power generation costs in 2020. Source: [81].

the Eq. (24).

$$CoE = \frac{PVC}{E_{out}} \tag{24}$$

$$PVC = C_{inv} + C_{omr} \left[\frac{1+I}{r-I}\right] \left[1 - \left(\frac{1+I}{1+r}\right)^n\right] - S\left(\frac{1+I}{1+r}\right)^n$$
(25)

$$PVB = UAB \frac{(1+DR)^n - 1}{DR(1+DR)^n}$$
(26)

$$DR = \frac{1+r}{1+I} - 1$$
 (27)

where UAB is the uniform annual benefit, which is mainly obtained using the selling CoE, n is the number of years, and DR is the discount rate. UAB is determined by using the energy output and purchase tariff of wind energy in Pakistan. Purchase tariff value of 7.4756 cent/kWh is taken in this study. DR is the discount rate given by Eq. (27) [85]. Beak-even analysis can be performed using Eq. (24)-(26), and the payback period for the wind turbine can be estimated.

V. HYDROGEN PRODUCTION

In this section, a system model is presented that can be used for generating Hydrogen from wind energy conversion system(WECS). The WECS utilizes double-fed induction generator (DFIG) wind turbines, AC-DC/DC-AC converters with their grid-side and rotor-side controllers (GSC and RSC), an electrolyzer, and a fuel-cell option. An embedded control system is responsible for controlling the DC bus voltage of the converter using a PID controller and its associated pulse width modulated (PWM) signal for a buck-boost converter.

The DC bus voltage can be further regulated using another DC–DC converter that supplies the desired DC voltage to the electrolyzer. The amount of Hydrogen produced depends on the energy yield from the wind turbine E_{out} , efficiency of the converter, η , and the energy consumed by the electrolyzer during the electrolysis process, E_{el} . The efficiency of a standard AC-DC converter ranges from 85% to 95%, and the energy consumed by electrolyzer E_{el} . varies between 5 and 6 kWh/Nm³. Assuming the efficiency of the electrolyzer is linearly proportional to the energy supplied and that all losses have been considered, Eq. (28) can be used to calculate the

amount of Hydrogen, aH_2 , produced [42], [87].

$$aH_2 = \frac{\eta \times E_{out}}{E_{el}} \tag{28}$$

The detailed dynamic model of WECS, electrolyzer, power electronics module, and the associated control is provided in [33]. One kilogram of Hydrogen is equal to 11.13 Nm³ [88]. The Hydrogen generated from different sources are dependent on many factors, such as the size of the wind turbine installed, and the weather profile that affects wind speeds. Because of the intermittent nature of wind energy, the DC bus voltage fed to the electrolyzer needs to be controlled. Therefore, the operational life of the electrolyzer is considered to be seven years in this study, and the results are correspondingly generated for short-term, midterm, and long-term Hydrogen production A good speed of wind, along with excess power, can be fed better to the electrolyzer. When wind speeds are low or there are no winds at all, a fuel-cell system can enhance the reliability of the system. The proposed scheme is shown in Fig. 5 and its associated power flow controller is shown in Fig. 6.

A. COST OF ELECTROLYZER

The most crucial component of the integrated hybrid system for Hydrogen production is the electrolyzer, whose cost has three main parts, e.g., capital cost, operation and maintenance (O&M) cost, and replacement cost. The capital cost is affected by the amount of Hydrogen produced, electrolyzer efficiency, CF, and the electrolyzer unit cost. The annual O&M cost and replacement costs are assumed to be 2% and 25% of the initial investment cost. The cost of the electrolyzer, *C*_{electrolyzer} is formulated as follows [89],

$$C_{electrolyzer} = \frac{aH_2 \times E_{th,elec} \times C_{electrolyzer,pu}}{8760 \times CF \times \varphi}$$
(29)

where $E_{th,elec}$ is the theoretical specific energy consumed by the electrolyzer assumed to be 5 kWh/Nm³ (or 55.6 kWh/kg), $C_{electrolyzer,pu}$ is the unit cost of the electrolyzer, which is taken as 384 USD/kWe in accordance with the target values mentioned in [84], CF is the capacity factor, and φ is the electrolyzer efficiency, which is approximately 75% for an alkaline electrolyzer.

B. COST OF HYDROGEN PRODUCTION

This cost of Hydrogen cH_2 is formulated as follows:

$$cH_2 = \frac{C_{electrolyzer} + CoE}{aH_2 \times n} \tag{30}$$

where $C_{electrolyzer}$ is the capital, operating, and maintenance cost of the electrolyzer; CoE is the cost of energy in USD; aH2 (kg/yr) is the annual Hydrogen production; and n represents the project lifetime. For long-term investment that would result in lower cost of Hydrogen production, it is economically feasible to install large-scale wind farms on the selected sites. DFIG based WECS coupled with electrolyzers and fuel cells for back-up, can be utilized in such an arrangement.

VI. DISCUSSION ON RESULTS

A. WIND RESOURCE ASSESSMENT

The Wind Resource Assessment has been done over wind speed data collected by World Bank under Energy Sector Management Assistance Program (ESMAP)] [90], [91] project for three locations selected in this study. Data set contains diurnal wind speed with 10-min intervals, direction of wind and atmospheric conditions say temperature, humidity and pressure. From this data, the mean wind speeds on monthly basis were calculated at four heights of 20m, 40m, 60m and 80m.

1) MONTHLY MEAN WIND SPEEDS

The mean wind speeds calculated on monthly basis for the selected sites are presented in Fig. 7(a) to Fig. 7(c).

Fig. 7(a) represents the monthly mean wind speed for Bahawalpur site for a time span of more than two years i.e., April 2016 to June 2018. It is evident that there is a seasonal wind speed variation. During a time span from March to August, a trend of higher wind speeds occurs with peaks in April and August. The peaks of average monthly wind speed at heights of 20m, 40m, 60m and 80m are 4.61, 5.29, 5.75, and 6.10 m/s, respectively. The annual average wind speeds for four heights are 3.57, 4.35, 4.83, and 5.16 m/s, respectively. During a span of September to February, a trend of lower wind speeds occurs with minimum values of 2.54, 3.18, 3.49, and 3.63 m/s, respectively for the four heights. With the wind-power classification given in Fig. 2(a), this site can be considered as marginal to fair, with a wind-power class performance of two or higher.

Fig. 7(b) represents the monthly mean wind speed for Sanghar site for a time span of more than two years i.e., June 2016 to October 2017. During a time span from April to August, a trend of higher wind speed occurs. The peaks of average monthly wind speed at heights of 20m, 40m, 60m and 80m were 7.32, 8.04, 8.6, and 9.08 m/s, respectively and the annual average wind speeds were 4.91, 5.68, 6.28, and 6.65 m/s, respectively. During a span of October to January, a trend of lower wind speed occurs with minimum values were 3.01, 3.83, 4.33, and 4.59 m/s for the four heights. With the wind-power classification given in Fig. 2(a), this site can be considered as fair to good with a class designation between three and four.

Fig. 7(c) represents the monthly mean wind speed for Gwadar site for a time span of more than two years i.e., November 2016 to October 2017. During a time span from March to June, a trend of higher wind speed occurs. The peaks of average monthly wind speed at heights of 20m, 40m, 60m and 80m were 4.87, 5.46, 5.74, and 5.95 m/s, respectively and the annual average wind speeds were 5.57, 6.50, 7.12, and 7.56 m/s, respectively. During a span of June to September, a trend of lower wind speed occurs with minimum values were 3.09, 3.63, 3.92, and 4.10 m/s for the four heights. With the wind-power classification given in Fig. 2(a), this site can be considered as marginal with class designation between two or higher.



FIGURE 5. Implementation strategy for wind-to-Hydrogen production. An integrated WECS coupled with an electrolyzer and a fuel cell through power electronic modules.

2) DIURNAL WIND SPEEDS

The mean wind speeds calculated on daily basis for the selected sites are presented in Fig. 8(a) through Fig. 8(c). As given in Fig. 8(a), the daily variations for Bahawalpur site lies between 3.26 to 5.87 m/s at heights of 20m, 40m, 60m and 80m. The wind speeds decreased at 1 AM till noon and then started to increase until midnight. For Sanghar and Gwadar, different wind speed patterns were observed, as shown in Fig. 8(b) and Fig. 8(c). These patterns show that Sanghar has a trend of higher wind speed, while Gwadar shows peaks at 11AM to 1PM.

3) ESTIMATED WEIBULL PDF AND CDF

The estimated Weibull PDFs and recorded wind speed data were analyzed for Bahawalpur, Sanghar and Gawadar for fitness and shown in Fig. 9 through Fig. 11 at heights of 20m, 40m, 60m and 80m. The Weibull parameters are estimated by four different techniques. From the figures, it is evident that at a height of 20m, the wind frequency probabilities are equal to 18%, 15%, and 17% at 5, 5.5, and 2.5 m/s wind speeds for Bahawalpur, Sanghar, and Gwadar respectively. The maximum wind probabilities varies at 40m by 17%, 14%, and 16% with 5, 5.25, and 2.5 m/s wind speeds, at 60m by 18%, 14%, and 16% with 5, 7.10 and 3.5 m/s wind speed and at 80m by 13.5%, 13%, and 15% with 5, 6.25, and 2.5 m/s wind speeds.

Next, the cumulative distribution function (CDF) of the corresponding Weibull PDFs are shown in Fig. 12 for these

sites at heights mentioned before. This CDF is another useful statistical indicator in describing the probabilities of wind speeds as far as cut-off and cut-in wind speed.

4) WIND ROSE DIAGRAMS OF SELECTED SITES

Installation of wind farms depends on the wind direction at that particular site. The wind direction and mean wind speed help in generating the wind frequency rose (%) diagram. The wind directions were recorded by sensors installed at heights of 58.5m and 78.5m and the corresponding wind rose plots, presented in [83], [84], are reproduced here. Fig. 13 through Fig. 15 represents the wind rose plots of the selected sites. These rose diagrams indicate the direction in which particular wind speed blows (north = N, east = E, south = S, west = W) with the frequency of occurrence. The prevailing wind directions for Bahawalpur are N, S, SSW, and NNW (Fig. 13), that for Sanghar is W (Fig. 14), and those for Gwadar are WSW and SSW (Fig. 15). The same wind directions also carry the most energy content.

5) ESTIMATED K AND C WEIBULL PARAMETERS

The estimated Weibull parameters namely k and c with their values at four different heights for the selected sites are given in the Table 5. The results from EM, EPF, and MLE are closely matched with each other. However, the results of the GM differ from all three methods mentioned earlier. At a height of 80m, Table 5 showed that the representative k and



FIGURE 6. Flow chart of the power controller of the proposed system.

c values using MLE were 1.92 and 6.16 for Bahawalpur, 2.12 and 7.34 for Sanghar, and 1.84 and 5.61 for Gwadar, respectively. Similarly, the values at heights of 20m, 40m and 60m depending on site and estimation method can be taken from Table 5.

6) GOODNESS OF FIT OF STATISTICAL PARAMETERS FOR SELECTED SITES

The statistical indicators for goodness of Fit are mentioned in Table 6. These indicators are used to compare the measured wind power and the calculated wind power from Weibull parameters using EM, EPF and MLE methods. The most frequent indicators for goodness of Fit are RMSE, RRMSE, MAPE, and correlation coefficient (R). Lowest RRMSE values are achieved with EM, EPF and MLE methods which represents the best accuracy and therefore better performance. However when GM method is used to calculate the Weibull parameters, based on RRMSE value the performance is worst. In case of R the best match between measured and estimated values occur at unity. Comparing to MLE, the EPF method has the lowest MAPE values at heights of 40m and 60m. MLE showed better performance than the other two methods because it achieved the lowest values for MAPE for Sanghar site where turbine installations at 40m, 60m, and 80m are more suitable. Table 6 clearly shows that GM has the highest RMSE values, whereas the other three methods show better error statistics (small error values). GM is not included for the Diebold-Mariano (DM) test due to its poor performance [85]. We performed the DM test for the remaining three estimator models (EM, EPF, and MLE); the results are tabulated in Table 7. We used the estimated PDFs of the fitted methods to obtain the error statistics given in Eq. (19)



FIGURE 7. Monthly mean wind speed with various hub heights at a) Bahawalpur b) Sanghar and c) Gwadar sites.

and used the square error loss function from Eq. (17) to calculate the values of the DM test. The result showed that PDF closely follows Weibull statistics. When compared with the true Weibull function obtained from the histogram data, DM test can be applied to verify if the fits are statistically significantly different.

According to the DM test results for Bahawalpur, at 60m, based on the square error loss function, DM values, i.e., 10.624, 12.97, and 17.83, are greater than the $Z_{\alpha/2}$ value of 1.96. Thus, hypothesis H_0 can be rejected at the 5% level of significance. This means that the observed differences between the forecasting performances of all three methods for a height of 60m for Bahawalpur are significantly different. Therefore, hypothesis H_1 is accepted. The results for a height of 80m also show that the three methods used for Bahawalpur are statistically different; thus, hypothesis H_1 can be accepted at a 5% level of significance. The same argument can be used

FABLE 5. Estimated k and c parameter	rs of Weibull distribution for selected sites.

Site	Height	GM		E	М	E	PF	M	LE
		k	с	k	с	k	с	k	с
Bahawalpur	20m	2.07	4.14	1.89	4.42	1.86	4.41	1.84	4.40
	40m	2.48	4.92	2.07	5.18	2.04	5.18	2.05	5.18
	60m	2.27	5.22	2.32	5.19	2.37	5.19	2.29	5.18
	80m	2.16	5.84	1.97	6.18	1.99	6.18	1.92	6.16
Sanghar	20m	2.33	5.38	2.07	5.63	2.06	5.63	2.06	5.64
	40m	2.30	6.24	2.19	6.37	2.20	6.37	2.17	6.37
	60m	2.21	6.93	2.21	7.00	2.24	7.00	2.19	6.99
	80m	2.15	7.27	2.15	7.35	2.20	7.35	2.12	7.34
Gwadar	20m	1.84	4.43	1.70	4.74	1.70	4.74	1.69	4.75
	40m	1.96	4.94	1.84	5.20	1.84	5.20	1.82	5.20
	60m	1.98	5.19	1.88	5.41	1.89	5.41	1.87	5.41
	80m	1.98	5.33	1.87	5.61	1.87	5.61	1.84	5.61



FIGURE 8. Daily mean wind speed with various hub heights at a) Bahawalpur b) Sanghar and c) Gwadar sites.

for the Sanghar and Gwadar sites. Results for heights of 60m and 80m indicate that the three methods used are statistically different. Thus, hypothesis H_1 can be accepted at a 5% level of significance. The last column in Table 7 provides the statistical conclusions regarding accepting hypothesis H_1 for the three methods adopted to estimate the parameters for these sites.

7) POWER DENSITY AND ENERGY DENSITY

The measured and estimated values of wind-power density are mentioned in Table 8. In order to calculate the annual wind energy density, the power density must be multiplied by the total number of hours in a year i.e., 8760 hrs. From Table 8, it is evident that the actual and estimated values are quiet close. For all sites, the highest values of annual wind energy densities are at the height of 80m. Further, Table 8 illustrate that Sanghar has the highest power density compared to the other two sites. Power densities calculated in Table 8 are compared with wind-power classification shown in Fig. 2 and in Table 11. It is evident that Sanghar is relatively a good site, while Gwadar and Bahawalpur sites belong to marginal wind-power classes.

8) ANALYSIS OF COST OF ENERGY AND PAYBACK PERIODS For all calculations, the conversion rate considered was 1 USD = 168.19 Pakistani Rupee (on September 11, 2021). Capacity factors are calculated according to Eq. (22) using estimated k and c Weibull parameters given in Table 5. The annual energy yield is then calculated using Eq. (21) and calculated values are listed in Table 10. Considering the electricity costs given in Table 10 it is evident that cost varies from one site to another site.

If DeWind D4/48 is used, the lowest CoE is 37.03 USD/ MWh for Sanghar (Table 10). If DeWind D6 is used, then the cost is 40.56 USD/MWh for Sanghar (ref. to Table 10). Therefore, it is evident that DeWind D4/48 turbines are promising options for these sites. Selecting Nordex N90/2500 wind turbines for Sanghar is a better option than DeWind D4/48. The CoE for Nordex N90/2500 is 35.21 USD/MWh for Sanghar. Payback periods can be calculated from Fig. 16 and the values are listed in Table 9. Using 2500 kW Nordex N90 could be more economical for wind-farm installation due to its higher CF of 25.73% among all sites. Another reason is that the payback period of Nordex N90/2500 is smaller (6 years) compared with that of DeWind D6, which is 7 years. Nordex N90/2500 is more economically beneficial for its lifetime and will result in less CoE per MWh. Irfan Ullah in [91] and Shahnawaz Farhan Khahro in [21] suggested that Nordex N90/2300 can be selected for installation in Gharo and Keti Bandar areas of Sindh in Pakistan. Therefore, all economic indicators suggest that Nordex N90/2500 is a good choice for the sites selected in our current work. DeWind D4/48 and Nordex N90/2500 are therefore, suitable wind turbines for installation of wind farms.

IEEE Access



FIGURE 9. Estimated PDF for Bahawalpur (a) 20m (b) 40m (c) 60m and (d) 80m heights.

TABLE 6. Statistical error indicators estimated for selected sites.

Site	Height		RM	ISE			MA	PE			RRM	ISE			i	R	
		GM	EM	EPF	MLE	GM	EM	EPF	MLE	GM	EM	EPF	MLE	GM	EM	EPF	MLE
Bahawalpur	20m	20.73	1.62	0.68	0.18	27.09	2.12	.89	0.24	$7.34 \times$	$4.48 \times$	$7.94 \times$	$5.84 \times$	0.93	0.99	0.99	1
							0			10^{-2}	10^{-4}	10^{-5}	10^{-6}				
	40m	30.51	2.03	0.43	0.97	27.41	1.82	0.38	0.87	$7.51 \times$	$43.34 \times$	$1.51 \times$	$7.70 \times$	0.921	0.99	1	0.99
										10^{-2}	10^{-4}	10^{-5}	10^{-6}				
	60m	4.86	1.37	0.21	1.81	4.95	1.40	0.21	1.84	$2.46 \times$	$1.96 \times$	$4.6 \times$	$3.4 \times$	0.99	0.99	1	0.99
										10^{-3}	10^{-4}	10^{-6}	10^{-4}				
	80m	44.40	0.46	1.64	4.20	22.78	0.23	0.84	2.15	$5.19 \times$	$5.71 \times$	$7.141 \times$	$4.67 \times$	0.94	1	0.99	0.99
										10^{-2}	10^{-6}	10^{-5}	10^{-4}				
Sanghar	20m	31.46	1.61	0.94	0.19	22.16	1.14	0.66	0.13	$4.92 \times$	$5.80 \times$	$1.89 \times$	$5.76 \times$	0.95	0.99	1	1
										10^{-2}	10^{-3}	10^{-5}	10^{-3}				
	40m	19.17	0.39	1.15	1.17	9.92	0.20	0.59	0.60	$9.85 \times$	$4.16 \times$	$3.58 \times$	$3.68 \times$	0.991	1	1	1
										10^{-3}	10^{-6}	10^{-5}	10^{-5}				
	60m	5.01	2.52	0.35	3.43	1.99	1	0.13	1.36	$3.99 \times$	$1.01 \times$	$1.95 \times$	$1.86 \times$	0.99	0.99	1	0.99
										10^{-4}	10^{-4}	10^{-6}	10^{-4}				
	80m	4	5.72	0.34	8.40	1.35	1.93	0.11	2.84	$1.84\times$	$3.75 \times$	$1.39\times$	$1.86 \times$	0.99	0.99	1	0.99
										10^{-4}	10^{-4}	10^{-6}	10^{-4}				
Gwadar	20m	28.48	0.20	0.20	1.39	26.77	0.19	0.19	1.31	$7.17 \times$	$3.70\times$	$3.70\times$	$1.72\times$	0.92	1	1	0.99
										10^{-2}	10^{-6}	10^{-6}	10^{-4}				
	40m	26.23	0.55	0.55	1.19	20.73	0.43	0.43	0.94	$4.30\times$	$1.93 \times$	$1.93 \times$	$8.88 \times$	0.95	1	1	0.99
										10^{-2}	10^{-5}	10^{-5}	10^{-5}				
	60m	23.05	0.10	0.76	0.98	16.69	0.07	0.55	0.71	$2.79\times$	$5.30 \times$	$3.10 \times$	$5.13 \times$	0.97	1	1	0.99
										10^{-2}	10^{-7}	10^{-5}	10^{-5}				
	80m	31.11	0.63	0.63	2.47	19.98	0.40	0.40	1.59	$3.99\times$	$1.66 \times$	$1.66 \times$	$2.53\times$	0.96	1	1	0.99
										10^{-2}	10^{-5}	10^{-5}	10^{-3}				



FIGURE 10. Estimated PDF for Sanghar (a) 20m (b) 40m (c) 60m and (d) 80m heights.

TABLE 7. DM test results.

Site	Height	DM for EMP and EPF	DM for EMP and MLE	DM for EPF and MLE	Results at 5% significance level
Bahawalpur	60m	10.62	12.97	17.83	Accept H_1 , Reject H_0
Bahawalpur	80m	31.34	30.54	30.82	Accept H_1 , Reject H_0
Sanghar	60m	15.86	-9.45	17.31	Accept H_1 , Reject H_0
Sanghar	80m	22.76	24.36	23.37	Accept H_1 , Reject H_0
Gwadar	60m	4.14	5.56	4.52	Accept H_1 , Reject H_0
Gwadar	80m	16.03	17.05	16.85	Accept H_1 , Reject H_0

The highest supported capacity factors are 24.50% from DeWind D4/48, 22.30% from DeWind D6, and 25.73% from Nordex N90/2500 for the Sanghar site.

B. ECONOMICS OF HYDROGEN PRODUCTION

1) YEARLY HYDROGEN PRODUCTION

Table 12 lists the values of annual Hydrogen production in (Ton-H2/Yr). The Hydrogen production was estimated from Eq. (28) using the annual energy estimate from Eq. (23). The efficiency of the converter, η , was taken as 90%, and the energy consumed by electrolyzer, E_{el} , was considered to be 5 kWh/Nm³. From the listed values of H_2 , it is clear that the

most promising site is Sanghar with a Hydrogen production of 91.14 ton/Yr. Clearly, if the Nordex N90/2500 wind turbine is used, the highest amount of Hydrogen can be generated on an annual basis. The corresponding values of Hydrogen production from Bahawalpur and Gwadar were 68.50 and 58.23 ton/yr, respectively.

2) COST OF HYDROGEN PRODUCTION

The cost of Hydrogen production was calculated for longterm, midterm, and short-term periods. Results in Table 12 reveal that for long-term investment, the minimum cost was achieved at Sanghar site with a corresponding cost of 2.29 k\$/ton using Nordex N90/2500 turbine and

IEEE Access



FIGURE 11. Estimated PDF for Gwadar (a) 20m (b) 40m (c) 60m and (d) 80m heights.



FIGURE 12. CDF of (a) Bahawalpur (c) Gwadar (c) Sanghar.

2.41 k\$/ton using DeWind D4/48 wind turbine. For midterm investment, the corresponding cost values for this site are 4.01 and 4.22 k\$/ton, respectively.

3) HYDROGEN DELIVERY OPTIONS FROM PRODUCTION SITES

Hydrogen produced can be utilized on-site or can be transported to demand sites for domestic usage or to meet industrial energy needs through a distribution network. The lack of existence and widespread of the fuel cell technology, electric cars, and Hydrogen fuel require extensive distribution infrastructure which needs to be developed. Generally, Hydrogen is distributed through three major modes of delivery:

1) Pipeline: This is most efficient and convenient method of large volume of Hydrogen delivery; however,



FIGURE 13. Wind Rose of Bahawalpur (a) Speed rose at 80 m hub height and direction sensor at 78.5 m height (b)Energy Rose at 80m and direction sensor at 78.5 m height.



FIGURE 14. Wind Rose of Sanghar (a) Speed Rose at 80m hub height and direction sensor at 78.5m height (b) Energy Rose at 80m and direction sensor at 78.5m height.



FIGURE 15. Wind Rose of Gwadar(a) Speed Rose at 80m hub height and direction sensor at 78.5m height (b) Energy Rose with at 80m and direction sensor at 78.5m height.

it requires huge capital investment and the length of the pipe impose restriction to the capacity of the Hydrogen being delivered.

- High-Pressure Trailers: For short distance and larger volume, Hydrogen in compressed gaseous state is considered ideal choice. However, it is relatively more expensive due to large cost associated with Hydrogen compression.
- 3) Liquid Hydrogen Tanks: Lowering the temperature of the liquid Hydrogen to cryogenic condition converts the gaseous Hydrogen to liquid. Although, this process is more expensive than others, and only recommend for large distances transportation of Hydrogen.

For the above three modes of transportation, it is essential to ensure that the consumption of Hydrogen and delivery of Hydrogen are matched with reasonable accuracy. For liquid

		Me	asured			Estimated					
Site	Height			(GM		EM	EPF		N N	1LE
		PD	AED	PD	AED	PD	AED	PD	AED	PD	AED
Bahawalpur	20m	76.51	670.23	55.78	488.63	74.89	656.04	75.83	664.27	76.33	668.65
	40m	111.31	975.08	80.80	707.81	109.27	957.21	110.88	971.31	110.33	966.49
	60m	98.12	859.53	102.98	902.10	99.49	957.21	97.90	857.60	99.93	875.39
	80m	194.85	1706.89	150.44	1317.85	195.31	1710.92	193.20	1692.43	199.05	1743.68
Sanghar	20m	141.91	1243.13	110.45	967.54	140.29	1228.94	140.97	1234.90	141.72	1241.47
_	40m	193.26	1692.96	174.08	1524.94	192.86	1689.45	192.10	1682.80	194.43	1703.21
	60m	251.41	2202.35	246.39	2158.38	253.93	2224.43	251.06	2199.29	254.84	2232.40
	80m	295.46	2588.23	291.45	2553.10	301.18	2638.34	295.11	2585.16	305.86	2665.81
Gwadar	20m	106.39	931.98	77.90	682.40	106.18	930.14	106.18	930.14	107.78	944.15
	40m	126.55	1108.58	100.31	878.72	125.99	1103.67	125.99	1103.67	127.74	1119.00
	60m	138.10	1209.76	115.05	1007.84	138.20	1210.63	137.33	1203.01	139.09	1218.43
	80m	155.73	1364.19	124.61	1091.58	155.10	1358.68	155.10	1358.68	158.21	1385.92

TABLE 8. Measured and estimated power density (PD) (W/m^2) and annual energy density (AED) (kWh/m^2) from the four selected methods.



FIGURE 16. Payback period using (a) Dewind D4 (b) Dewind D6 (c) Nordex N90/2500.

TABLE 9. Pa	yback	period	of wir	id turbines.
-------------	-------	--------	--------	--------------

Turbine	Payback Period (years)			
DeWind D4/48	6			
DeWind D6	7			
Nordex N90/2500	6			

Hydrogen, it is critically important to use it at high rate of release at the utilization site else it evaporates from the storage vessel. Therefore, the Liquid Organic Hydrogen Carriers (LOHC) are commonly used in the liquid Hydrogen transportation, LOHC absorbs and release the Hydrogen through chemical reaction. LOHC can be transported over long distances [92].

The Hydrogen distribution and delivery infrastructure to link the distribution sites with the fueling stations or point of utilization is complex and complicated task which presents a number of challenges. Due to lower energy density of the Hydrogen, it is very expensive to store, transport and deliver Hydrogen compared with other fuels. Due to unique properties of Hydrogen, the storage and delivery equipment is relatively costly, and thus results in significant increase in Hydrogen delivery cost. Considering the fact that there are number of various resources and technologies to produce Hydrogen, the centralized and distributed Hydrogen production can counteract the cost of transportation and delivery.

nomically viable option of Hydrogen transportation is either as liquid Hydrogen through trucks, or as gaseous Hydrogen in trailers. For transportation and utilization of the Hydrogen, the closest cities from the Hydrogen production sites have been selected, as presented in Table 13. In order to simplify the transportation route, a transportation radius based on longest distance has been chosen. Therefore, the Hydrogen produce at Sanghar will be transported within a radius of 250km, at Bahawalpur within a radius of 450 km, and at Gwadar within a radius of 650 km. The costing model for Hydrogen transportation is based on [93], under similar transportation settings and assumption. The Hydrogen transportation cost include the capital cost (vehicle, compressor etc.), labor, and vehicle fuel cost. The Hydrogen transportation cost for each site to their transportation radius is presented in Table 14. Supply-chain model for Hydrogen production, storage and transport network can be visualized through Fig. 17.

Considering the huge investment cost of pipeline, the eco-

Since all relevant costs have been considered, therefore the results of the Hydrogen transportation cost are presented in USD/kg for specific radius of each side as shown in Table 14. If the transportation radius is changed, the cost of the Hydrogen transportation will also increase due to additional cost of the vehicle fuel. The results also have reasonable agreement with the Hydrogen transportation cost estimated in number

TABLE 10. Capacity factor, annual energy production (MWh/Yr), and cost of Energy(\$/MWh).

Turbine	Site	CF	E(MWh/Yr)	CoE(\$/MWh) (n=20 Yr)
DeWind D4/48	Bahawalpur	0.168	16998	56.03
	Sanghar	0.245	25722	37.03
	Gwadar	0.189	19930	47.79
DeWind D6	Bahawalpur	0.100	21900	90.60
	Sanghar	0.223	48925	40.56
	Gwadar	0.148	32324	61.38
Nordex N90/2500	Bahawalpur	0.193	84709	46.85
	Sanghar	0.257	112697	35.21
	Gwadar	0.164	72007	55.11



FIGURE 17. Supply-chain model for Hydrogen production, storage and transport network.

of other studies [94], [95]. It can be seen that the liquid Hydrogen is more feasible for larger distances, generally, for transportation distances of more than 500km [96]. For distances under 500km, the gaseous Hydrogen transportation is economically feasible than liquid Hydrogen. The compressor energy usage, and equipment cost for the liquid transportation is expensive than gaseous Hydrogen. Sanghar site has the lowest transportation cost with the gaseous liquid as economically viable mode of transportation. The transportation of the Hydrogen from Gwadar is economically not feasible, as the cost of transportation either as liquid or gas phase is above 17 USD/kg, almost three times the production cost of the Hydrogen.

The sites are usually geographically displaced and efficient transportation of this fuel is a major concern. In this

TABLE 11. Wind power classification for sites.

Site	Wind Power Class	Class Designator
Bahawalpur	Marginal to Fair	2-3
Sanghar	Fair to Good	3-4
Gwadar	Marginal	2

regards many approaches are possible for example consume Hydrogen where it is produced thus meeting the local demand of the locality. If Hydrogen supply is surplus then it can be dispatched to other distribution centers. With the development of smart cities in near future and increased penetration of renewable energy, the flux of electric vehicles (EVs) with batteries are steadily increasing. Hydrogen fuel is an alternative solution to EV which if utilized can

Turbine	Site	$H_2(ton/Yr)$	$\begin{array}{c} CoH_2(k\$/ton) \\ (n=7 \mathrm{Yr}) \end{array}$	$\begin{array}{c} CoH_2(k\$/ton) \\ (n=4 \ \mathrm{Yr}) \end{array}$	$\begin{array}{c} CoH_2(k\$/ton) \\ (n=1 \text{ Yr}) \end{array}$
DeWind D4/48	Bahawalpur	17.34	3.65	6.39	25.55
DeWind D4/48	Sanghar	20.80	2.41	4.22	16.88
DeWind D4/48	Gwadar	16.11	3.11	5.45	21.79
DeWind D6	Bahawalpur	17.42	5.90	10.33	41.32
DeWind D6	Sanghar	41.15	2.64	4.62	18.48
DeWind D6	Gwadar	27.19	4.00	7.00	27.98
Nordex N9/2500	Bahawalpur	68.50	3.05	5.34	21.35
Nordex N9/2500	Sanghar	91.14	2.29	4.01	16.04
Nordex N9/2500	Gwadar	58.23	3.59	6.28	25.11

TABLE 12. Cost of hydrogen for long-term(n = 7 Yr), mid-term(n = 4 Yr) and short-term(n = 1 Yr) investment.

TABLE 13. H_2 production sites and distance of closest cities.

	Distance of closest cities		
H_2 production sites	1	2	3
Sanghar	Nawabshah - 65 km	Hyderabad - 120 km	Karachi - 260 km
Bahawalpur	Multan - 105 km	Faisalabad - 335 km	Lahore - 430 km
Gwadar	Pasni - 126 km	Turbat - 160 km	Karachi - 631 km

TABLE 14. H₂ transportation cost for a given transport radius from production site.

	Transportation radius	Transportation cost	
H_2 production sites		Gaseous H_2	Liquid H_2
Sanghar	250 km	7.52 USD/kg	17.2 USD/kg
Bahawalpur	450 km	14.2 USD/kg	19.3 USD/kg
Gwadar	650 km	17.6 USD/kg	21.1 USD/kg

further reduce pollution in the environment and create new job market. There are still many hurdles in totally moving towards Hydrogen economy the major factors being low profit, less societal adaption and higher logistics, operational and maintenance costs. In the near future when fossil fuels will deplete or the supply becomes expensive then demand for EVs and Hydrogen based vehicles integrated with fuel cells will increase.

For energy efficiency and optimizing cost of supply-chain network, an efficient Hydrogen dispatch model is required. In this model Hydrogen tank refilling and swapping stations play a key role which will directly affect operational cost of the network. A coordinated effort is required to provide logistics support for multi-regions operation and management. Costs for storage and compression of Hydrogen are also important considerations in overall cost optimization process. It has been reported in [97] that demand for Hydrogen fuel will increase and by year 2050 many countries of the world will be enjoying Hydrogen as a fuel. The source of this Hydrogen production will be mostly from renewable energy sources such as wind and solar. The number of Hydrogen re-filling and swapping stations will be increased because of increased Hydrogen demand. Even the day-ahead or month-ahead forecast would become inevitable to meet the Hydrogen demand market. Hydrogen re-fill station (HRS) and Hydrogen swapping stations (HSS) are geographically displaced, so a logistic system is required to keep track of Hydrogen tanks needed in a particular location. HRS will be responsible for Hydrogen re-filling and



FIGURE 18. Multi-region Hydrogen tank filling and swapping network.

HSS will be responsible for Hydrogen tank swapping at stations. A Multi-region Hydrogen Tank Filling and Swapping



FIGURE 19. Production and possible distribution sites for Hydrogen.

Network (MHTFSN) will coordinate all supply-chains and will provide the required coordination among various regions.

Due to increased penetration of information and communication technology (ICT), MHFTSN will interact with regional supply-chains of Hydrogen swapping and refilling stations (HSRS) which will provide services within their communities. The need for efficient and cost effective distribution network will arise. Therefore these HSRS will work under the control of distribution system operators (DSO). In our current work we have proposed such a supply-chain model for Hydrogen demand and transport network as shown in Fig. 18. However due to different consumer's Hydrogen consumption patterns, the prices for Hydrogen transport may vary. The market clearing prices (MCP) may change dramatically in open energy market and the stations can participate in Hydrogen fuel trading [98]. DSOs will help in this energy trading. Based on the wind power forecast using artificial neural networks (ANN) or machine learning approaches for a region and MCP forecast using statistical models, a multi-objective optimization algorithm can be developed for Hydrogen production schedule for a day-ahead or month-ahead energy market and optimize production cost of a particular station. The algorithm can also optimize the transportation cost, storage and compression cost subject to certain constraints by transporting either compressed Hydrogen [99] or liquid organic Hydrogen [100].

Based on our finding for feasible sites for Hydrogen production with lowest cost values, production sites as well as distribution nodes can be identified. These sites are identified in distribution network for Hydrogen as shown in Fig. 19. From the map it is clear that Gwadar site can serve Pasni and Turbat, whereas Sanghar site can meet the Hydrogen demand of Hyderabad and Nawabshah. However, Karachi is at a distance of 260km and therefore transport of Hydrogen is not much feasible. Wind sites close to Karachi such as Gharo (at a distance of 86km), Jhimpir (at a distance of 127.3km via Hyderabad Motorway), Keti Bandar (at a distance of 154.1 km via Gharo Rd/Gharo-Keti Bunder Hwy/N-110), Babaurband village (about 67 km in Northeast of Karachi, Sindh) are better suited for production of Hydrogen. Province Sindh, identified with its rich wind corridors for energy production can benefit from this study and move towards Hydrogen economy by the end of 2035 if Hydrogen production plants are installed. Surplus Hydrogen can be transported to other major cities such as Bahawalpur, Multan, Faisalabad, Lahore. This will move Pakistan towards a real Hydrogen economy with socio-economic benefits for the society and drastically reducing oil and gas imports.

VII. CONCLUSION

In this work, the wind resource potential was probed for three sites located in Punjab, Sindh, and Baluchistan provinces of Pakistan. The actual data of wind speeds at the selected sites between November 2016 to October 2017 were analyzed to determine the mean wind speeds on monthly and daily basis for the selected sites suggesting that these sites belong to a class performance of two or higher wind potential. The parameters of the Weibull distribution were estimated by GM, EM, EPF, and MLE methods. The wind turbine models with the highest capacity factors were proposed to install wind farms at mentioned sites. The actual wind profile data recorded from selected sites tightly fit the Weibull distribution function. The scale parameters k and c for 80m installations were 2.12 and 7.34, respectively, for Sanghar. The obtained values were validated using statistical error indicators. Further, the wind-power density and annual energy yield were estimated using Weibull parameters. DeWind D4/48 and Nordex N 90/2500 wind turbines were recommended for wind-to-Hydrogen production. The highest capacity factor values were 24.47% from DeWind D4/48, 22.34% from DeWind D6, and 25.73% from Nordex N90/2500 for Sanghar. Concerning cost of energy, Sanghar can be regarded as a suitable site for wind energy projects followed by Bahawalpur and Gwadar.

The amount of Hydrogen production, using coupled wind energy conversion system with water electrolyzer was estimated for Bahawalpur, Sanghar and Gwadar sites. The cost of Hydrogen production was also estimated. An implementation scheme was provided for practical realization of wind-to-Hydrogen production. In terms of Hydrogen production using electrolysis, Sanghar is the best site. The cost of Hydrogen was calculated for three different periodslong-term (7 years), midterm (4 years), and short-term periods (1 year). For long-term investment, the minimum cost was achieved at Sanghar with the corresponding cost of Hydrogen being 2.29 USD/kg using Nordex N90/2500. Transportation cost of Hydrogen was also determined based on a given transport radius. Hydrogen is considered as one of the major energy carriers for future energy systems and transition to renewable and zero-carbon energy technologies is inevitable. This paper particularly aimed to analyze the techno-economic aspect of Hydrogen production from available wind resources in Pakistan. Transportation cost is relatively higher compared to the Hydrogen production cost. It is found that the cost of Hydrogen transportation is 1.2 to 4 times higher than the cost of Hydrogen production. Including the detailed costing and economics of modes and ways of Hydrogen transportation is multi-variable, complex, complicated and challenging optimization task and therefore will be considered in our future work.

REFERENCES

- A. Gupta, "Climate change and Kyoto protocol: An overview," in Handbook of Environmental and Sustainable Finance, V. Ramiah and G. N. Gregoriou, Eds. San Diego, CA, USA: Academic, 2016, ch. 1, pp. 3–23.
- [2] E. Maibach, J. Miller, F. Armstrong, O. El Omrani, Y. Zhang, N. Philpott, S. Atkinson, L. Rudoph, J. Karliner, J. Wang, C. Pétrin-Desrosiers, A. Stauffer, and G. K. Jensen, "Health professionals, the Paris agreement, and the fierce urgency of now," *J. Climate Change Health*, vol. 1, Mar. 2021, Art. no. 100002.
- [3] D. G. Caglayan, H. U. Heinrichs, M. Robinius, and D. Stolten, "Robust design of a future 100% renewable European energy supply system with hydrogen infrastructure," *Int. J. Hydrogen Energy*, vol. 46, no. 57, pp. 29376–29390, Aug. 2021.
- [4] M. S. Nazir, M. Bilal, H. M. Sohail, B. Liu, W. Chen, and H. M. N. Iqbal, "Impacts of renewable energy atlas: Reaping the benefits of renewables and biodiversity threats," *Int. J. Hydrogen Energy*, vol. 45, no. 41, pp. 22113–22124, Aug. 2020.
- [5] A. Crivellari and V. Cozzani, "Offshore renewable energy exploitation strategies in remote areas by power-to-gas and power-to-liquid conversion," *Int. J. Hydrogen Energy*, vol. 45, no. 4, pp. 2936–2953, Jan. 2020.
- [6] S. Hossain, H. Chowdhury, T. Chowdhury, J. U. Ahamed, R. Saidur, S. M. Sait, and M. A. Rosen, "Energy, exergy and sustainability analyses of Bangladesh's power generation sector," *Energy Rep.*, vol. 6, pp. 868–878, Nov. 2020.

- [8] Y.-S. Huang and S.-J. Liu, "Chinese green hydrogen production potential development: A provincial case study," *IEEE Access*, vol. 8, pp. 171968–171976, 2020.
- [9] F. Chien, H. W. Kamran, G. Albashar, and W. Iqbal, "Dynamic planning, conversion, and management strategy of different renewable energy sources: A sustainable solution for severe energy crises in emerging economies," *Int. J. Hydrogen Energy*, vol. 46, no. 11, pp. 7745–7758, Feb. 2021.
- [10] S. A. A. Shah, "Feasibility study of renewable energy sources for developing the hydrogen economy in Pakistan," *Int. J. Hydrogen Energy*, vol. 45, no. 32, pp. 15841–15854, Jun. 2020.
- [11] (2021). Renewable Capacity Statistics 2021 International Renewable Energy Agency Irena (2021). [Online]. Available: https://www.irena.org/ publications/2021/March/Renewable-Capacity-Statistics-2021
- [12] (2021). BP Statistical Review of World Energy 2021, 70th Edition. [Online]. Available: https://www.bp.com/content/dam/bp/businesssites/en/global/corporate/pdfs/energy-economics/ statistical-review/bp-stats-review-2021-full-report.pdf
- [13] M. A. Saeed, Z. Ahmed, S. Hussain, and W. Zhang, "Wind resource assessment and economic analysis for wind energy development in Pakistan," *Sustain. Energy Technol. Assessments*, vol. 44, Apr. 2021, Art. no. 101068.
- [14] M. A. Saeed, Z. Ahmed, and W. Zhang, "Optimal approach for wind resource assessment using Kolmogorov–Smirnov statistic: A case study for large-scale wind farm in Pakistan," *Renew. Energy*, vol. 168, pp. 1229–1248, May 2021.
- [15] M. H. Baloch, G. S. Kaloi, and Z. A. Memon, "Current scenario of the wind energy in Pakistan challenges and future perspectives: A case study," *Energy Rep.*, vol. 2, pp. 201–210, Nov. 2016.
- [16] H. Saulat, M. M. Khan, M. Aslam, M. Chawla, S. Rafiq, F. Zafar, M. M. Khan, A. Bokhari, F. Jamil, A. W. Bhutto, and A. A. Bazmi, "Wind speed pattern data and wind energy potential in Pakistan: Current status, challenging platforms and innovative prospects," *Environ. Sci. Pollut. Res.*, vol. 28, no. 26, pp. 34051–34073, Jul. 2021.
- [17] M. Sumair, T. Aized, S. A. R. Gardezi, M. M. A. Bhutta, S. M. S. Rehman, and S. U. Rehman, "Comparison of three probability distributions and techno-economic analysis of wind energy production along the coastal belt of Pakistan," *Energy Explor. Exploitation*, pp. 1–23, Jun. 2020.
- [18] Z. H. Hulio, W. Jiang, and S. Rehman, "Techno–economic assessment of wind power potential of Hawke's bay using Weibull parameter: A review," *Energy Strategy Rev.*, vol. 26, Nov. 2019, Art. no. 100375.
- [19] M. Gul, N. Tai, W. Huang, M. Nadeem, and M. Yu, "Assessment of wind power potential and economic analysis at Hyderabad in Pakistan: Powering to local communities using wind power," *Sustainability*, vol. 11, no. 5, p. 1391, Mar. 2019.
- [20] S. F. Khahro, K. Tabbassum, A. M. Soomro, L. Dong, and X. Liao, "Evaluation of wind power production prospective and Weibull parameter estimation methods for babaurband, Sindh Pakistan," *Energy Convers. Manage.*, vol. 78, pp. 956–967, Feb. 2014.
- [21] S. F. Khahro, K. Tabbassum, A. Mahmood Soomro, X. Liao, M. B. Alvi, L. Dong, and M. F. Manzoor, "Techno-economical evaluation of wind energy potential and analysis of power generation from wind at Gharo, Sindh Pakistan," *Renew. Sustain. Energy Rev.*, vol. 35, pp. 460–474, Jul. 2014.
- [22] M. M. Aman, G. B. Jasmon, A. Ghufran, A. H. A. Bakar, and H. Mokhlis, "Investigating possible wind energy potential to meet the power shortage in Karachi," *Renew. Sustain. Energy Rev.*, vol. 18, pp. 528–542, Feb. 2013.
- [23] I. A. Mirza, N. A. Khan, and N. Memon, "Development of benchmark wind speed for Gharo and Jhimpir, Pakistan," *Renew. Energy*, vol. 35, no. 3, pp. 576–582, Mar. 2010.
- [24] I. A. Gondal, S. A. Masood, and R. Khan, "Green hydrogen production potential for developing a hydrogen economy in Pakistan," *Int. J. Hydrogen Energy*, vol. 43, no. 12, pp. 6011–6039, Mar. 2018.
- [25] A. Mayyas, M. Wei, and G. Levis, "Hydrogen as a long-term, large-scale energy storage solution when coupled with renewable energy sources or grids with dynamic electricity pricing schemes," *Int. J. Hydrogen Energy*, vol. 45, no. 33, pp. 16311–16325, Jun. 2020.

- [26] J. Carton and A. Olabi, "Wind/hydrogen hybrid systems: Opportunity for Ireland's wind resource to provide consistent sustainable energy supply," *Energy*, vol. 35, no. 12, pp. 4536–4544, Dec. 2010.
- [27] A. Rahil, R. Gammon, N. Brown, J. Udie, and M. U. Mazhar, "Potential economic benefits of carbon dioxide (CO₂) reduction due to renewable energy and electrolytic hydrogen fuel deployment under current and long term forecasting of the social carbon cost (SCC)," *Energy Rep.*, vol. 5, pp. 602–618, Nov. 2019.
- [28] K. Bareiß, C. de la Rua, M. Möckl, and T. Hamacher, "Life cycle assessment of hydrogen from proton exchange membrane water electrolysis in future energy systems," *Appl. Energy*, vol. 237, pp. 862–872, Mar. 2019.
- [29] P. Burmistrz, T. Chmielniak, L. Czepirski, and M. Gazda-Grzywacz, "Carbon footprint of the hydrogen production process utilizing subbituminous coal and lignite gasification," *J. Cleaner Prod.*, vol. 139, pp. 858–865, Dec. 2016.
- [30] U. K. Mirza, N. Ahmad, K. Harijan, and T. Majeed, "A vision for hydrogen economy in Pakistan," *Renew. Sustain. Energy Rev.*, vol. 13, no. 5, pp. 1111–1115, Jun. 2009.
- [31] A. Mohammadi and M. Mehrpooya, "A comprehensive review on coupling different types of electrolyzer to renewable energy sources," *Energy*, vol. 158, pp. 632–655, Sep. 2018.
- [32] B. Olateju, J. Monds, and A. Kumar, "Large scale hydrogen production from wind energy for the upgrading of bitumen from oil sands," *Appl. Energy*, vol. 118, pp. 48–56, Apr. 2014.
- [33] O. Alavi, A. Mostafaeipour, and M. Qolipour, "Analysis of hydrogen production from wind energy in the southeast of Iran," *Int. J. Hydrogen Energy*, vol. 41, no. 34, pp. 15158–15171, Sep. 2016.
- [34] A. Al-Sharafi, A. Z. Sahin, T. Ayar, and B. S. Yilbas, "Techno-economic analysis and optimization of solar and wind energy systems for power generation and hydrogen production in Saudi Arabia," *Renew. Sustain. Energy Rev.*, vol. 69, pp. 33–49, Mar. 2017.
- [35] A. Mostafaeipour, M. Khayyami, A. Sedaghat, K. Mohammadi, S. Shamshirband, M.-A. Sehati, and E. Gorakifard, "Evaluating the wind energy potential for hydrogen production: A case study," *Int. J. Hydrogen Energy*, vol. 41, no. 15, pp. 6200–6210, Apr. 2016.
- [36] M. Rezaei-Shouroki, A. Mostafaeipour, and M. Qolipour, "Prioritizing of wind farm locations for hydrogen production: A case study," *Int. J. Hydrogen Energy*, vol. 42, no. 15, pp. 9500–9510, Apr. 2017.
- [37] Z. N. Ashrafi, M. Ghasemian, M. I. Shahrestani, E. Khodabandeh, and A. Sedaghat, "Evaluation of hydrogen production from harvesting wind energy at high altitudes in Iran by three extrapolating Weibull methods," *Int. J. Hydrogen Energy*, vol. 43, no. 6, pp. 3110–3132, Feb. 2018.
- [38] I. Dincer and C. Acar, "Smart energy solutions with hydrogen options," *Int. J. Hydrogen Energy*, vol. 43, no. 18, pp. 8579–8599, May 2018.
- [39] M. Gökçek and C. Kale, "Techno-economical evaluation of a hydrogen refuelling station powered by wind-PV hybrid power system: A case study for İzmir-Çeşme," *Int. J. Hydrogen Energy*, vol. 43, no. 23, pp. 10615–10625, Jun. 2018.
- [40] T. R. Ayodele and J. L. Munda, "Potential and economic viability of green hydrogen production by water electrolysis using wind energy resources in South Africa," *Int. J. Hydrogen Energy*, vol. 44, no. 33, pp. 17669–17687, Jul. 2019.
- [41] D. Chade, T. Miklis, and D. Dvorak, "Feasibility study of wind-tohydrogen system for Arctic remote locations–Grimsey island case study," *Renew. Energy*, vol. 76, pp. 204–211, Apr. 2015.
- [42] M. Douak and N. Settou, "Estimation of hydrogen production using wind energy in Algeria," *Energy Proc.*, vol. 74, pp. 981–990, Aug. 2015.
- [43] F. Sayedin, A. Maroufmashat, S. Sattari, A. Elkamel, and M. Fowler, "Optimization of photovoltaic electrolyzer hybrid systems; taking into account the effect of climate conditions," *Energy Convers. Manage.*, vol. 118, pp. 438–449, Jun. 2016.
- [44] S. Koumi Ngoh, L. M. Ayina Ohandja, A. Kemajou, and L. Monkam, "Design and simulation of hybrid solar high-temperature hydrogen production system using both solar photovoltaic and thermal energy," *Sustain. Energy Technol. Assessments*, vol. 7, pp. 279–293, Sep. 2014.
- [45] S. Koumi Ngoh and D. Njomo, "An overview of hydrogen gas production from solar energy," *Renew. Sustain. Energy Rev.*, vol. 16, no. 9, pp. 6782–6792, Dec. 2012.
- [46] M. R. Shaner, H. A. Atwater, N. S. Lewis, and E. W. McFarland, "A comparative technoeconomic analysis of renewable hydrogen production using solar energy," *Energy Environ. Sci.*, vol. 9, no. 7, pp. 2354–2371, 2016.

- [47] M. S. Genç, M. Çelik, and I. Karasu, "A review on wind energy and windhydrogen production in Turkey: A case study of hydrogen production via electrolysis system supplied by wind energy conversion system in central Anatolian Turkey," *Renew. Sustain. Energy Rev.*, vol. 16, no. 9, pp. 6631–6646, Dec. 2012.
- [48] G. Genç, M. Çelik, and M. S. Genç, "Cost analysis of wind-electrolyzerfuel cell system for energy demand in Pinarbaşi-Kayseri," *Int. J. Hydrogen Energy*, vol. 37, no. 17, pp. 12158–12166, Sep. 2012.
- [49] P. Menanteau, M. M. Quéméré, A. Le Duigou, and S. Le Bastard, "An economic analysis of the production of hydrogen from windgenerated electricity for use in transport applications," *Energy Policy*, vol. 39, no. 5, pp. 2957–2965, May 2011.
- [50] B. Olateju, A. Kumar, and M. Secanell, "A techno-economic assessment of large scale wind-hydrogen production with energy storage in Western Canada," *Int. J. Hydrogen Energy*, vol. 41, no. 21, pp. 8755–8776, Jun. 2016.
- [51] S. Rahimi, M. Meratizaman, S. Monadizadeh, and M. Amidpour, "Techno-economic analysis of wind turbine–PEM (polymer electrolyte membrane) fuel cell hybrid system in standalone area," *Energy*, vol. 67, pp. 381–396, Apr. 2014.
- [52] J. Sigurvinsson, C. Mansilla, P. Lovera, and F. Werkoff, "Can high temperature steam electrolysis function with geothermal heat?" *Int. J. Hydrogen Energy*, vol. 32, no. 9, pp. 1174–1182, Jun. 2007.
- [53] S. Rahmouni, N. Settou, N. Chennouf, B. Negrou, and M. Houari, "A technical, economic and environmental analysis of combining geothermal energy with carbon sequestration for hydrogen production," *Energy Proc.*, vol. 50, pp. 263–269, Jan. 2014.
- [54] W. Iqbal, H. Yumei, Q. Abbas, M. Hafeez, M. Mohsin, A. Fatima, M. Jamali, M. Jamali, A. Siyal, and N. Sohail, "Assessment of wind energy potential for the production of renewable hydrogen in Sindh Province of Pakistan," *Processes*, vol. 7, no. 4, p. 196, Apr. 2019.
- [55] S. Siddique and R. Wazir, "A review of the wind power developments in Pakistan," *Renew. Sustain. Energy Rev.*, vol. 57, pp. 351–361, May 2016.
- [56] T. P. Chang, "Performance comparison of six numerical methods in estimating Weibull parameters for wind energy application," *Appl. Energy*, vol. 88, pp. 272–282, Jan. 2011.
- [57] J. A. Carta, P. Ramírez, and S. Velázquez, "Influence of the level of fit of a density probability function to wind-speed data on the WECS mean power output estimation," *Energy Convers. Manage.*, vol. 49, no. 10, pp. 2647–2655, Oct. 2008.
- [58] S. A. Akdag and A. Dinler, "A new method to estimate Weibull parameters for wind energy applications," *Energy Convers. Manage.*, vol. 50, pp. 1761–1766, Jul. 2009.
- [59] Global Wind Atlas. Accessed: Aug. 28, 2021. [Online]. Available: https://globalwindatlas.info/area/Pakistan/
- [60] M. Adnan, J. Ahmad, S. F. Ali, and M. Imran, "A techno-economic analysis for power generation through wind energy: A case study of Pakistan," *Energy Rep.*, vol. 7, pp. 1424–1443, Nov. 2021.
- [61] J. Wang, X. Huang, Q. Li, and X. Ma, "Comparison of seven methods for determining the optimal statistical distribution parameters: A case study of wind energy assessment in the large-scale wind farms of China," *Energy*, vol. 164, pp. 432–448, Dec. 2018.
- [62] M. H. Baloch, D. Ishak, S. T. Chaudary, B. Ali, A. A. Memon, and T. A. Jumani, "Wind power integration: An experimental investigation for powering local communities," *Energies*, vol. 12, no. 4, p. 621, Feb. 2019.
- [63] M. A. U. Haq, S. Chand, M. Z. Sajjad, and R. M. Usman, "Evaluating the suitability of two parametric wind speed distributions: A case study from Pakistan," *Model. Earth Syst. Environ.*, vol. 7, no. 3, pp. 1683–1691, Jul. 2020.
- [64] M. Khalid Saeed, A. Salam, A. U. Rehman, and M. Abid Saeed, "Comparison of six different methods of Weibull distribution for wind power assessment: A case study for a site in the northern region of Pakistan," *Sustain. Energy Technol. Assessments*, vol. 36, Dec. 2019, Art. no. 100541.
- [65] M. Teimouri, S. M. Hoseini, and S. Nadarajah, "Comparison of estimation methods for the Weibull distribution," *Statistics*, vol. 47, no. 1, pp. 93–109, Feb. 2013.
- [66] C. Carrillo, J. Cidrás, E. Díaz-Dorado, and A. Obando-Montaño, "An approach to determine the Weibull parameters for wind energy analysis: The case of Galicia (Spain)," *Energies*, vol. 7, no. 4, pp. 2676–2700, Apr. 2014.
- [67] W. Werapun, Y. Tirawanichakul, and J. Waewsak, "Comparative study of five methods to estimate Weibull parameters for wind speed on Phangan island, Thailand," *Energy Proc.*, vol. 79, pp. 976–981, Nov. 2015.

- [68] Y. A. Kaplan, "Determination of the best Weibull methods for wind power assessment in the southern region of Turkey," *IET Renew. Power Gener.*, vol. 11, no. 1, pp. 175–182, Sep. 2016.
- [69] C. G. Justus, R. Hargraves, A. Mikhail, and D. Graber, "Methods for estimating wind speed frequency distributions," *J. Appl. Meteorol.*, vol. 17, no. 3, pp. 350–353, 1977.
- [70] P. K. Chaurasiya, S. Ahmed, and V. Warudkar, "Study of different parameters estimation methods of Weibull distribution to determine wind power density using ground based Doppler SODAR instrument," *Alexandria Eng. J.*, vol. 57, no. 4, pp. 2299–2311, Dec. 2018.
- [71] Z. H. Hulio, W. Jiang, and S. Rehman, "Techno–economic assessment of wind power potential of Hawke's Bay using Weibull parameter: A review," *Energy Strategy Rev.*, vol. 26, Nov. 2019, Art. no. 100375.
- [72] T. Orłowska-Kowalska, F. Blaabjerg, and J. Rodríguez, Eds., Advanced and Intelligent Control in Power Electronics and Drives. Cham, Switzerland: Springer, 2014.
- [73] M. M. Alam, S. Rehman, J. P. Meyer, and L. M. Al-Hadhrami, "Review of 600–2500 kW sized wind turbines and optimization of hub height for maximum wind energy yield realization," *Renew. Sustain. Energy Rev.*, vol. 15, no. 8, pp. 3839–3849, Oct. 2011.
- [74] T. Haghroosta, "Comparative study on typhoon's wind speed prediction by a neural networks model and a hydrodynamical model," *MethodsX*, vol. 6, pp. 633–640, Mar. 2019.
- [75] X. Niu and J. Wang, "A combined model based on data preprocessing strategy and multi-objective optimization algorithm for short-term wind speed forecasting," *Appl. Energy*, vol. 241, pp. 519–539, May 2019.
- [76] K. Mohammadi, O. Alavi, A. Mostafaeipour, N. Goudarzi, and M. Jalilvand, "Assessing different parameters estimation methods of Weibull distribution to compute wind power density," *Energy Convers. Manage.*, vol. 108, pp. 322–335, Jan. 2016.
- [77] A. Tizpar, M. Satkin, M. Roshan, and Y. Armoudli, "Wind resource assessment and wind power potential of Mil-E Nader region in Sistan and Baluchestan Province, Iran–Part 1: Annual energy estimation," *Energy Convers. Manage.*, vol. 79, pp. 273–280, Mar. 2014.
- [78] O. Nematollahi, P. Alamdari, M. Jahangiri, A. Sedaghat, and A. A. Alemrajabi, "A techno-economical assessment of solar/wind resources and hydrogen production: A case study with GIS maps," *Energy*, vol. 175, pp. 914–930, May 2019.
- [79] F. Blaabjerg and K. Ma, "Future on power electronics for wind turbine systems," *IEEE J. Emerg. Sel. Topics Power Electron.*, vol. 1, no. 3, pp. 139–152, Sep. 2013.
- [80] Determination of New Tariff for Wind Power Generation Projects, National Electric Power Regulatory Authority Islamic Republic Pakistan, NEPRA, Islamabad, Pakistan, 2017.
- [81] Irena. (2021). Renewable Power Generation Costs in 2020, International Renewable Energy Agency, Abu Dhabi. Accessed: Jun. 2021. [Online]. Available: https://www.irena.org/publications/2021/Jun/Renewable-Power-Costs-in-2020
- [82] Determination of the National Electric Power Regulatory Authority in the Matier of Tariff Petition Filed by Iran-Pak Wind Power (pvt.) Limited for Determination of Generation Tariff in Respect of 50 MW Wind Power Project, National Electric Power Regulatory Authority Islamic Republic Pakistan, NEPRA, Islamabad, Pakistan, 2021.
- [83] J. Ahmad, M. Imran, A. Khalid, W. Iqbal, S. R. Ashraf, M. Adnan, S. F. Ali, and K. S. Khokhar, "Techno economic analysis of a windphotovoltaic-biomass hybrid renewable energy system for rural electrification: A case study of Kallar Kahar," *Energy*, vol. 148, pp. 208–234, Apr. 2018.
- [84] Pakistan–Economic Indicators, T. E. Pakistan, New York, NY, USA, 2019.
- [85] R. Ziazi, K. Mohammadi, and N. Goudarzi, "Techno-economic assessment of utilizing wind energy for hydrogen production through electrolysis," in *Proc. ASME Power Conf.*, vol. 57618, 2017, Art. no. V002T09A019.
- [86] R. B. Gupta, Hydrogen Fuel: Production, Transport, and Storage. Boca Raton, FL, USA: CRC Press, 2009.
- [87] M. J. Khan and M. T. Iqbal, "Analysis of a small wind-hydrogen stand-alone hybrid energy system," *Appl. Energy*, vol. 86, no. 11, pp. 2429–2442, Nov. 2009.
- [88] M. Rezaei, M. Salimi, M. Momeni, and A. Mostafaeipour, "Investigation of the socio-economic feasibility of installing wind turbines to produce hydrogen: Case study," *Int. J. Hydrogen Energy*, vol. 43, no. 52, pp. 23135–23147, Dec. 2018.

- [89] Pakistan–Wind Measurement Data, World Bank Group, Washington, DC, USA, 2018.
- [90] O. Yilidrimar and B. Adiloglu, "Wind resource mapping in Pakistan: Twelve month site resource report," The World Bank, Washington, DC, USA, Tech. Rep. 138076, 2018.
- [91] I. Ullah, Q.-U.-Z. Chaudhry, and A. J. Chipperfield, "An evaluation of wind energy potential at Kati Bandar, Pakistan," *Renew. Sustain. Energy Rev.*, vol. 14, no. 2, pp. 856–861, Feb. 2010.
- [92] M. Niermann, S. Timmerberg, S. Drünert, and M. Kaltschmitt, "Liquid organic hydrogen carriers and alternatives for international transport of renewable hydrogen," *Renew. Sustain. Energy Rev.*, vol. 135, Jan. 2021, Art. no. 110171.
- [93] Z. Luo, Y. Hu, H. Xu, D. Gao, and W. Li, "Cost-economic analysis of hydrogen for China's fuel cell transportation field," *Energies*, vol. 13, no. 24, p. 6522, Dec. 2020.
- [94] Y. Li and F. Taghizadeh-Hesary, "Energy prices and the economic feasibility of using hydrogen energy for road transport in the people's republic of China," Asian Develop. Bank Inst., Tokyo, Japan, Tech. Rep. 1185, 2020. [Online]. Available: https://www.adb. org/publications/energy-prices-economic-feasibility-hydrogen-energyroad-transport-prc
- [95] J. Ogden and C. Yang, "Implementing a hydrogen energy infrastructure: Storage options and system design," in *MRS Proceedings*. vol. 895. Davis, CA, USA: Univ. California, 2005.
- [96] M. Reu, T. Grube, M. Robinius, P. Preuster, P. Wasserscheid, and D. Stolten, "Seasonal storage and alternative carriers: A flexible hydrogen supply chain model," *Appl. Energy*, vol. 200, pp. 290–302, Aug. 2017.
- [97] A. Lahnaoui, C. Wulf, H. Heinrichs, and D. Dalmazzone, "Optimizing hydrogen transportation system for mobility by minimizing the cost of transportation via compressed gas truck in North Rhine-Westphalia," *Appl. Energy*, vol. 223, pp. 317–328, Aug. 2018.
- [98] W. Xiao, Y. Cheng, W.-J. Lee, V. Chen, and S. Charoensri, "Hydrogen filling station design for fuel cell vehicles," *IEEE Trans. Ind. Appl.*, vol. 47, no. 1, pp. 245–251, Jan./Feb. 2011.
- [99] A. Lahnaoui, C. Wulf, H. Heinrichs, and D. Dalmazzone, "Optimizing hydrogen transportation system for mobility via compressed hydrogen trucks," *Int. J. Hydrogen Energy*, vol. 44, no. 35, pp. 19302–19312, Jul. 2019.
- [100] C. Wulf, M. Reuß, T. Grube, P. Zapp, M. Robinius, J.-F. Hake, and D. Stolten, "Life cycle assessment of hydrogen transport and distribution options," *J. Cleaner Prod.*, vol. 199, pp. 431–443, Oct. 2018.



JAMEEL AHMAD received the B.Sc. degree in electrical engineering from UET Peshawar, in 1993, the M.S. degree in systems engineering from Quaid-i-Azam University Islamabad, Islamabad (CNS Campus), in 1996, the M.Sc. degree in electrical engineering from the University of Southern California (USC), Los Angeles, USA, in 2005, and the Ph.D. degree in electrical engineering from UET Lahore. He worked at PAEC, from 1996 to 2002, as a

Hardware Engineer. He worked at Qualcomm and Broadcom Corporation, San Diego, CA, USA, from June 2007 to June 2010, on 2G and 3G mobile communication systems. He also served for the Department of Electrical Engineering, SEN, University of Management and Technology (UMT), Lahore, Pakistan, for over ten years, and remained the Director for final year projects at the Department of BSEE for four years. He is currently working as an Assistant Professor with the Department of Computer Science, UMT. His research interests include distributed control and optimization of smart microgrids for energy internet, cloud computing, parallel and distributed computing, machine learning techniques applied to face recognition, energy, wireless systems, and connected and autonomous vehicles. He was a recipient of the MOST Scholarship from the Ministry of Science and Technology, Pakistan, from 2002 to 2006, and received the Best Researcher Award, in 2019, from UMT, for publishing a paper in IEEE TRANSACTIONS ON INDUSTRIAL INFORMATICS. He is a reviewer of numerous journals and conferences.



MUHAMMAD IMRAN received the Ph.D. degree in energy system engineering from the University of Science and Technology, South Korea. He is currently a Lecturer in mechanical and design engineering with Aston University, U.K. He is also an Established Researcher in the area of thermal energy systems. He was a part of the team at Korea Institute of Energy Research that successfully developed and commercialized 5, 30, and 100 kW ORC systems. His research activities

aimed at developing innovative thermal energy systems and improving the energy performance of existing energy systems. He has been working in the area of waste heat to power conversion technologies (organic Rankine cycle), since 2012. His work in the organic Rankine cycle technology can be categorized into design and optimization, working fluid selection, flow boiling and condensation of refrigerants, binary mixtures, volumetric expanders, system integration, and dynamic modeling and control. His research involves both the numerical modeling and experimental approach. He has the opportunity to conduct research in the above-mentioned research areas in Asia and Europe.



SYED REHAN ASHRAF received the bachelor's degree in industrial engineering and the master's degree in engineering management from the NED University of Engineering and Technology, Karachi, Pakistan, in 2006 and 2010, respectively. He worked as an Executive with the Department of Production and Quality, HinoPak Motors Ltd. (HPML), for six years. In August 2013, he joined the Department of Industrial Engineering, School of Engineering, University of Management and

Technology (UMT), Lahore, as an Assistant Professor, where he currently teaches courses of industrial and mechanical engineering. His research interests include the fields of optimization, operations management, and quality management. Further interests include cost minimization, reduction in energy input, and cost of quality. He has contributed to research on optimized usage of the fuel mix in cement plant and cost of quality. His ongoing research is on transportation models of goods from the collection point to the main plant. His most recent work is focused on operations of assembly and bus body plants, planning of trim line and quality control.



SYED FAROOQ ALI received the Ph.D. (CS) degree from UMT, Pakistan, and the Ph.D. course work, the Ph.D. comprehensive exam, and the M.S. (CS) degree from The Ohio State University, Columbus, USA, and the M.S. (CS) degree (Hons.) from LUMS, Lahore, Pakistan. During his stay in the M.S., he was involved on LUMS Fellowship. He is currently working as an Assistant Professor, UMT. His research interests include computer vision, digital image processing, and medical

imaging. He is a reviewer for various IEEE conferences and journals.



ZAWAR HUSSAIN received the bachelor's degree in electrical engineering from UMT, Lahore, in 2009, and the M.Sc. degree in electronics engineering with specialization in communication systems engineering from Hochschule Bremen University of Applied Sciences, Germany. During his academics, he was associated with several institutes as a Research Assistant, including the Laboratory of Optoelectronics and Light-wave Technology and the Institute of Aerospace & Tech-

nology, Hochschule Bremen. Moreover, he also worked with leading German automotive industries, including Siemens and BMW. He has been working as an Assistant Professor with the Department of Electrical Engineering, School of Engineering, UMT, since October 2014.



MUHAMMAD ADNAN received the bachelor's degree in electrical engineering (specialization in electronics and communication) from UCET, Bahuaddin Zakariya University, Multan, in 2002, the M.S. degree in telecommunication engineering from UET, Peshawar, in 2005, and the Ph.D. degree in information and communication systems from Beijing University of Posts and Telecommunications, China, in 2010. He served for PTCL-Etisalat Academy, from 2004 to 2010,

and the NFC Institute of Engineering and Technology, Multan, from 2011 to 2013. He has been a member of the Faculty of Electrical Engineering, UMT, since October 2013. His research interests include information and communication, energy, and artificial intelligence. He is a reviewer of numerous journals and conferences.



MUHAMMAD SHOAIB received the bachelor's degree in electrical engineering from the University of Engineering and Technology (UET) Lahore, Pakistan, and the master's degree from the University of Management and Technology (UMT), Lahore. He is currently working as an Assistant Professor with the School of Engineering (SEN), UMT. His research interests include energy and power, power systems, power electronics, and electric machines design.

•••